

Received 24 April 2024, accepted 12 May 2024, date of publication 14 May 2024, date of current version 22 May 2024.

Digital Object Identifier 10.1109/ACCESS.2024.3401078

THEORY

Graphs Drawn With Some Vertices per Face: Density and Relationships

CARLA BINUCCI^{®1}, GIUSEPPE DI BATTISTA^{®2}, WALTER DIDIMO^{®1}, VIDA DUJMOVIĆ^{®3}, SEOK-HEE HONG^{®4}, MICHAEL KAUFMANN^{®5}, GIUSEPPE LIOTTA^{®1}, (Senior Member, IEEE), PAT MORIN^{®6}, AND ALESSANDRA TAPPINI^{®1}

¹Department of Engineering, University of Perugia, 06123 Perugia, Italy

²Department of Civil, Computer Science and Aeronautical Technologies Engineering, University "Roma Tre", 00154 Roma, Italy

³School of Computer Science and Electrical Engineering, University of Ottawa, Ottawa, ON K1N 6N5, Canada

⁴School of Computer Science, The University of Sydney, Sydney, NSW 2050, Australia

⁵Wilhelm-Schickard Institut für Informatik, University of Tübingen, 72076 Tübingen, Germany

⁶School of Computer Science, Carleton University, Ottawa, ON K1S 5B6, Canada

Corresponding author: Alessandra Tappini (alessandra.tappini@unipg.it)

This work was supported in part the Italian Ministry for Universities and Research (MUR) Research Projects of National Relevance (PRIN)—"EXPAND: scalable algorithms for EXPloratory Analyses of heterogeneous and dynamic Networked Data" under Project 2022TS4Y3N; and in part by the Department of Engineering, University of Perugia, Ricerca di Base, "Modelli, algoritmi e sistemi per la visualizzazione e l'analisi di grafi e reti" under Grant RICBA22CB. This research started at the Summer Workshop on Graph Drawing (SWGD) 2022.

ABSTRACT Graph drawing beyond planarity is a research area that has received an increasing attention in the last twenty years, driven by the necessity to mitigate the visual complexity inherent in geometric representations of non-planar graphs. This research area stems from the study of graph layouts with forbidden crossing configurations, a well-established subject in geometric and topological graph theory. In this context, the contribution of this paper is as follows: 1) We introduce a new hierarchy of graph families, called k^+ real face graphs; for any integer k > 1, a graph G is a k^+ -real face graph if it admits a drawing Γ in the plane such that the boundary of each face (formed by vertices, crossings, and edges) contains at least kvertices of G (" k^+ " stands for k or more); 2) We give tight upper bounds on the edge density of k^+ -real face graphs, namely we prove that n-vertex 1⁺-real face and 2⁺-real face graphs have at most 5n - 10 and 4n - 8 edges, respectively. Furthermore, in a constrained scenario in which all vertices must lie on the boundary of the external face, 1⁺-real face and 2⁺-real face graphs have at most 3n - 6 and 2.5n - 4 edges, respectively; 3) We characterize the complete graphs that admit a k^+ -real face drawing or an outer k^+ -real face drawing for any $k \ge 1$. We also provide a clear picture for the majority of complete bipartite graphs; and 4) We establish relationships between k^+ -real face graphs and other prominent beyond-planar graph families; notably, we show that for any $k \ge 1$, the class of k⁺-real face graphs is not included in any family of beyond-planar graphs with hereditary property.

INDEX TERMS Beyond-planar graph drawing, edge density, geometric graph theory, graph visualization, inclusion relationships.

I. INTRODUCTION

Graph drawing (also known as graph visualization) is a well-established research area that addresses the problem of automatically computing readable visual representations of graphs and networks, by developing models, algorithms,

The associate editor coordinating the review of this manuscript and approving it for publication was Abdullah Iliyasu^D.

and systems [2], [3], [4]. Graph drawing finds application in many real-world domains where data are naturally modeled as complex graphs and the user can benefit from visualization paradigms and tools for exploring the data space. Examples of these domains include social sciences, software engineering, biology, finance, and computer networks (see, e.g., [5], [6], [7], [8], [9]). The importance of graph visualization has also been highlighted in the contexts of machine learning, knowledge discovery, and explainable AI, which further characterizes the interdisciplinary nature of this research area (see, e.g., [10], [11], [12], [13]).

Within the above scenario, the study of non-planar drawings of graphs with forbidden substructures has a long tradition, stemming from topological and geometric graph theory (see, e.g., [14]). This topic is commonly recognized as *beyond-planar graph drawing*, and it has become increasingly popular in the last two decades. From an application perspective, this growth in interest is motivated by the necessity of visually representing complex relational data (modeled as non-planar graphs), by reducing the negative impact of edge crossings on the readability of a graph visualization. This motivation is further reinforced by some human cognitive experiments designed to assess users' analytical capabilities when specific types of edge crossings are avoided in graph visualizations. See [15], [16], [17], and [18] for surveys or books on the subject.

A beyond-planar graph family is a class of non-planar graphs that can be drawn in the plane by avoiding some specific type of edge crossing configurations. For instance, for a given positive integer k, the family of k-planar graphs contains all graphs that admit a drawing with no more than k crossings per edge [19], [20], while k-quasi planar graphs are those that can be drawn without k mutually (i.e., pairwise) crossing edges [21], [22], [23], [24], [25], [26]. A generalization of k-planar graphs, called min-kplanar graphs has been recently proposed, in which for any two crossing edges one of the two must contain at most k crossings [27], [28]. Other prominent examples of beyond-planar graph families are *fan-planar* graphs [29], [30], [31], [32], [33], [34], [35], where an edge is not allowed to cross two independent edges, and k-gap planar graphs $(k \ge 1)$ [36], [37], [38], where for each pair of crossing edges one of the two edges contains a small gap through which the other edge can pass, and only k gaps per edge are allowed. In other words, in a k-gap planar drawing it is possible to assign each crossing to one of the two edges that form it, in such a way that no more than k crossings are assigned to the same edge. Geometric properties of edge crossings have also been proposed. In particular, right-angle-crossing graphs (RAC graphs for short) are those graphs that admit a drawing with straight-line edges such that any two crossing edges form angles of 90° at their crossing point [39], [40]; generalizations and variants of RAC drawings consider the possibility of inserting bends along the edges or forbid crossing angles that are below a given threshold (see, e.g., [41], [42], [43], [44]). RAC graphs are among the most investigated families in beyond-planar graph drawing, because there are human cognitive studies showing that a good angular resolution at the crossing points makes a graph layout more readable [45]. Refer to [16] for additional references on the subject.

Given a family \mathcal{F} of beyond-planar graphs, a core problem with a long tradition, which originates from extremal graph theory [46], [47], [48], is establishing the *edge density* for the elements of \mathcal{F} , i.e., the maximum number of edges that an *n*-vertex graph in \mathcal{F} can have with respect to *n*. Besides its theoretical interest, finding upper bounds on the edge density of graphs in a family \mathcal{F} represents a key factor in the design of recognition algorithms for \mathcal{F} , i.e., algorithms that take as input a graph and establish whether this graph belongs to \mathcal{F} (see, e.g., [30], [33], [49], [50], [51]). For example, it is known that *n*-vertex 1-planar graphs and 2-planar graphs have at most 4n - 8 edges and 5n - 10 edges, respectively, and both these bounds are tight, in the sense that there are graphs in these families that can actually achieve them [20], [52]. In the literature, a graph of \mathcal{F} whose number of edges is the maximum possible over its number of vertices is usually called an *optimal* graph of \mathcal{F} (see, e.g., [49], [53], [54], [55], [56], [57]).

A complementary research direction investigates how different beyond-planar graph families relate to each other in terms of inclusion [16]. For instance, the family of simple *k*-planar graphs is a subset of (k+1)-quasi planar graphs for any $k \ge 2$ [58]. Other examples of inclusion relationships involve the families of fan-planar and *k*-gap planar graphs [33], [37].

A. CONTRIBUTION

In this paper we introduce a new hierarchy of graph families. Namely, for any positive integer k, we define the family of k^+ -real face graphs as follows. Let Γ be a graph G in the plane, where edge crossing points are regarded as dummy vertices. The drawing Γ divides the plane into topologically connected regions, called *faces* (or *cells*): if Γ is a planar drawing (i.e., if Γ does not have crossings), the boundary of each face consists of vertices (and edges) of G; otherwise, there are some faces whose boundaries contain dummy vertices. We say that G is a k^+ -real face graph if it admits a drawing such that each face boundary contains at least k real vertices, i.e., k vertices of G ("k⁺" stands for "k or more"). Note that, for any $k \ge 1$, the family of $(k + 1)^+$ -real face graphs is included in the family of k^+ -real face graphs. For example, Figure 1 shows three different drawings of the same graph, namely the complete graph on five vertices. While the leftmost drawing does not belong to any family of k^+ -real face drawings (as it contains a face formed by crossing points only), the other two drawings are a 1^+ -real face and a 2^+ -real face drawing, respectively.

The study of the k^+ -real face graph hierarchy has both a theoretical and a practical motivation. From a theoretical perspective, k^+ -real face drawings are a generalization of planar drawings whose face sizes are above a desired threshold [59], [60], [61]. Also, finding k^+ -real face graphs can be regarded as a generalization to non-planar graphs of the classical guarding planar graph problem [62], where the vertices that cover the face set are the (real) vertices of *G*. From the practical side, the advantage of computing k^+ -real face drawings comes from the observation that faces mostly consisting of crossing points carry out little information about the relational dataset modeled by the graph and make the layout less readable. Indeed, the number of real vertices per face provides a measure of how much the drawing is far from



FIGURE 1. Three different drawings of the complete graph K_5 : (a) A drawing with a face that has no vertices (in gray); (b) a 1⁺-real face drawing; (c) a 2⁺-real face drawing.

TABLE 1. Summary of density results in this paper; n, m, and χ denote the number of vertices, edges, and crossings, respectively.

Graph Family	Crossings ($\chi \leq$)	Edges ($m \leq$)	Ref.
k^+ -real face graphs ($k \ge 3$)	$\frac{2-k}{k} \cdot m + n - 2$	$\frac{k}{k-2}(n-2)$	Le. 1,Th. 1
2 ⁺ -real face graphs	n-2	4n - 8	Le. 1,Th. 2
1 ⁺ -real face graphs	m+n-2	5n - 10	Le. 1,Th. 3
outer k^+ -real face graphs ($k \ge 3$)	$\frac{2-k}{k} \cdot m + \frac{k-1}{k} \cdot n - 1$	$\frac{k-1}{k-2} \cdot n - \frac{k}{k-2}$	Le. 2,Th. 4
outer 2 ⁺ -real face graphs	$\frac{1}{2}n - 1$	2.5n - 4	Le. 2, Th. 5
outer 1 ⁺ -real face graphs	m-1	3n - 6	Le. 2, Th. 6

being planar; when possible, faces formed only by crossing points should be avoided.

It is also worth remarking that our study differs from all previous approaches in the field of graph drawing beyond planarity. Indeed, the class of k^+ -real face graphs focuses on relating edge crossings to the structure of face boundaries, rather than just considering forbidden crossing configurations along edges. Our results are as follows:

- We provide tight upper bounds on the edge density of k^+ -real face graphs, for all values of k, both in the general case and in the constrained scenario in which all vertices of the graph are forced to stay on the external face. The constrained scenario can be seen as a generalization of the study of outerplanar graphs to our graph family, which we call *outer* k^+ -*real face graphs*. We remark that similar constraints have been previously studied for other families of beyond-planar graphs (see, e.g., [30], [33], [51], [63], [64]). Table 1 reports the set of our results on edge density. See also Table 2 for a comparison between our results and previous results in graph drawing beyond planarity.
- We use the edge density results to characterize the complete graphs that admit k^+ -real face drawings, for any integer $k \ge 1$. Interestingly, we observe the existence of k^+ -real face drawings of complete graphs that match both the maximum value of k and the minimum number of edge crossings, i.e., the so-called *crossing number* of the graphs. We also provide a clear picture for the majority of complete bipartite graphs.
- We establish inclusion relationships between *k*⁺-real face graphs and families of beyond-planar graphs with hereditary property, such as *h*-planar and *h*-quasi planar graphs. Notably, we show that, for any positive integer *k*,

the family of k^+ -real face graphs is not included in any beyond-planar graph family with hereditary property. However, we prove that this is not always the case if we restrict our attention to optimal graphs.

We remark that very recently, and after the conference version of the present paper was published [1], Kaufmann et al. [65] published a technical report about a new general formula, used to provide alternative proofs for several edge density results in the beyond-planar graph drawing area. In particular, the authors of [65] describe an alternative proof of the upper bounds on the edge density of unconstrained k^+ -real face graphs given in our paper. The work in [65] confirms the correctness of our results and reinforces the scientific interest in the graph families introduced in our article.

B. PAPER STRUCTURE

The remainder of the paper is structured as follows: Section II discusses the literature that is mainly related to our contribution; Section III contains basic notation and terminology adopted throughout the paper; Sections IV and V provide the edge-density results of k^+ -real face and outer k^+ -real face graphs, respectively; Section VI focuses on k^+ real face drawings of complete graphs and complete bipartite graphs; Section VII presents the results about inclusion relationships of k^+ -real face graphs and other beyond-planar graph families; Section VIII concludes with final remarks and suggestions about future research directions.

II. RELATED WORK

Our work provides advances on two main lines of research in graph drawing beyond planarity, namely the study of the edge density of specific graph families and the study **TABLE 2.** Edge density bounds for some popular beyond-planar graph families; *n* and *m* denote the number of vertices and edges of the graphs. The bounds assume simple graphs and drawings.

Graph Family	Edges ($m \leq$)	Tight	Ref.
1-planar	4n - 8	•	[52]
2-planar	5n - 10	•	[52]
3-planar	5.5n - 11	+	[66]
4-planar	6n - 12	+	[67]
k -planar ($k \ge 5$)	$3.81\sqrt{k}n$	×	[67]
3-quasi planar	6.5n - 20	•	[22], [65]
4-quasi planar	72(n-2)	×	[21]
k -quasi planar ($k \ge 5$)	$c_k n \log n$	0	[26]
fan-planar	5n - 10	•	[35], [65]
1-gap planar	5n - 10	•	[38]
k -gap planar ($k \ge 1$)	$O(\sqrt{k}n)$	×	[38]
0-bend RAC	4n - 10	•	[40]
1-bend RAC	5n - 10	•	[65]
2-bend RAC	10n - 19	+	[65]

of inclusion relationships among different graph families. In the following we briefly survey the literature on these two research lines, and we also discuss other common topics in the field.

A. EDGE DENSITY

The problem of establishing the maximum number of edges that an *n*-vertex graph in a given beyond-planar graph family can have is extensively studied in the literature. Notably, the majority of beyond-planar graph classes that have been investigated exhibit linear upper bounds on the edge density. Nonetheless, unlike testing graph planarity, deciding whether an input graph belongs to some given beyond-planar graph family is typically an NP-hard problem. While the interested reader can refer to the survey in [16] for the edge density results of a wide range of graph families, we summarize in Table 2 the bounds for some of the most popular graph families. For each result, we report the corresponding reference. Also, the symbol • indicates that the bound is tight; the symbols + and \times indicate that the bound is tight up to an additive or to a multiplicative constant, respectively; the symbol \circ means that the bound may be far from being tight.

The definition of most of the families reported in Table 2 has already been given in Section I. The last three lines in the table refer to straight-line RAC drawings (0-bend RAC), RAC drawings with at most one bend per edge (1)-bend RAC), and RAC drawings with at most two bends per edge (2)-bend RAC). We also recall that every graph admits a RAC drawing with at most three bends per edge, i.e, *n*-vertex 3-bend RAC drawings can achieve the density of the complete graph K_n [40].

B. INCLUSION RELATIONSHIPS

Table 2 shows that the maximum edge density of different beyond-planar graph families is sometimes the same. In these cases it is natural asking whether the two families coincide. In fact, this is not true most of the times. For example, although both 2-planar graphs and fan-planar graphs can have up to 5n - 10 edges, it is proven that none of these classes is included in the other [34].

There are also cases in which two families are not in any inclusion relationship, even if the maximum edge density of one of the two families is strictly less than the maximum edge density of the other. To this regard, a remarkable example is given by 0-bend RAC graphs (whose density is at most 4n-10), which are not included in the class of 1-planar graphs (whose density can be higher, namely up to 4n - 8) [68]. As another example, there exist 1-gap planar graphs that are not *k*-planar, for any given $k \ge 1$ [38]. Recall that *k*-planar graphs might have much higher density than 1-gap planar graphs when $k \ge 3$ (see Table 2).

A non-trivial inclusion relationship involves the hierarchies of k-planar graphs and k-quasi planar graphs. Namely, by using sophisticated rerouting arguments, Angelini et al. [58] showed that every k-planar drawing can be transformed into a (k + 1)-quasi planar drawing, for any integer $k \ge 2$. This implies that the family of k-planar graphs is properly included in the family of k-quasi planar graphs. Another interesting result by Bae et al. [38] establishes that all 2k-planar graphs are k-gap planar, for any $k \ge 1$; this result is proven by exploiting the famous Hall's theorem [69].

Refer to [16] for other relationships between beyond-planar graph families.

C. OTHER TOPICS IN BEYOND-PLANAR GRAPH DRAWING Although our paper concentrates on edge density and inclusion relationships for k^+ -real face graphs, it is worth remarking that other types of problems are studied in the literature in the field of beyond-planar graph drawing. One of them is the already mentioned *recognition problem*, that aims to establish the membership of graphs to specific graph families. Another problem concerns the stretchability of topological drawings, that is, establishing whether a certain type of beyond-planar drawing can be transformed into a drawing of the same family using straight-line edges only. While this is always true for planar graphs with fixed embedding, as the famous Fary's theorem proves [70], there are examples of beyond-planar drawings that are not stretchable, or that are stretchable only if we can change the embedding [71], [72], [73], [74].

To conclude this short literature review, we observe that many of the classical problems arising in beyond-planar graph drawing are addressed in constrained scenarios. As mentioned in Section I, some of our density results are about the constrained scenario in which all vertices belong to the external face. Other common constrained settings assume that the vertices are drawn on two parallel lines (e.g., when the graph is bipartite) or that all vertices lie on the same line (called *spine*) and the edges are drawn on a given number of distinct planes that contain this line (called *pages*). See [16] for an extensive discussion of this and other topics on graph drawing beyond planarity.

TABLE 3. Symbols for the cardinality of sets of vertices, edges, and faces.

V(G)	E(G)	$ V(\Gamma) $	$ E(\Gamma) $	$ F(\Gamma) $	$ V(\Gamma) ackslash V(G) $
n	т	ν	μ	φ	χ

III. BASIC NOTATION AND TERMINOLOGY

Let G be a graph. We assume that G is simple, that is, it contains neither multiple edges (also called parallel edges) nor self-loops. We also assume, without loss of generality, that G is connected, as otherwise we can just consider each connected component of G independently. We denote by V(G) and E(G) the set of vertices and the set of edges of G, respectively. A *drawing* Γ of G is a geometric representation of G that maps each vertex $v \in V(G)$ to a distinct point of the plane and each edge $uv \in E(G)$ to a simple Jordan arc between the points corresponding to *u* and *v*. We assume that Γ is a *simple* drawing (sometimes called *good* drawing in the literature), that is: (i) adjacent edges do not intersect, except at their common endpoint; (ii) two independent (i.e., nonadjacent) edges intersect at most in one of their interior points, called a crossing point; and (iii) no three edges intersect at a common crossing point.

Refer to Figure 2 for an illustration of the next definitions. Let Γ be a drawing of G. A vertex of Γ is either a point corresponding to a vertex of G, called a real-vertex, or a point corresponding to a crossing point, called a *crossing-vertex*. Observe that a crossing-vertex has degree four. We remark that in the literature a plane graph obtained by replacing crossing points with dummy vertices is often referred to as a *planarization* [3]. We denote by $V(\Gamma)$ the set of vertices of Γ . An *edge* of Γ is a curve connecting two vertices of Γ ; an edge of Γ whose endpoints are both real-vertices coincides with an edge of G. We denote by $E(\Gamma)$ the set of edges of Γ . Drawing Γ subdivides the plane into topologically connected regions, called *faces* (sometimes called *cells* in the literature). The boundary of each face consists of a circular sequence of vertices and edges of Γ . The set of faces of Γ is denoted by $F(\Gamma)$. Exactly one face in $F(\Gamma)$ corresponds to an infinite region of the plane, called the *external face* (or *outer face*) of Γ (face f_e in Figure 2(a)); the other faces are the *internal* faces of Γ . When the boundary of a face f of Γ contains a vertex v (or an edge e), we also say that f contains v (or e).

From now on, we denote by n = |V(G)| and m = |E(G)| the number of vertices and the number of edges of *G*, respectively. For a drawing Γ of *G*, we denote by $\nu = |V(\Gamma)|$, $\mu = |E(\Gamma)|$, and $\varphi = |F(\Gamma)|$ the number of vertices, edges, and faces of Γ , respectively. Also, we denote by $\chi = |V(\Gamma) \setminus V(G)| = \nu - n$ the number of crossing-vertices of Γ . For example in Figure 2(b) the number of crossing-vertices of Γ_2 is $\chi = 7$. Table 3 summarized the symbols used to denote the cardinality of the different vertex, edge, and face sets. Also, refer to Table 5 at the end of the paper for a more comprehensive glossary of the symbols used in the technical parts.



FIGURE 2. Two non-planar drawings Γ_1 and Γ_2 of the same graph *G* having n = 13 vertices and m = 22 edges. In both drawings the real-vertices are represented by empty circles, while the crossing vertices are represented by black circles. (a) The number of vertices of Γ_1 is $\nu = 21$ (it has 8 crossing-vertices). The number of edges and the number of faces of Γ_1 are $\mu = 38$ and $\varphi = 19$, respectively. Face f_1 is a 0-real face, face f_4 is a 2-real triangle while face f_2 and face f_3 are a 1-real quadrilateral and a 2-real quadrilateral, respectively. Face f_e is the external face. For face f_2 we have that $\delta_{\Gamma}^r(f_2) = 1$, $\delta_{\Gamma}^c(f_2) = 3$, and then $\delta_{\Gamma}(f_2) = 4$. The boundary of face f_5 is not a simple cycle since vertex *z* is traversed twice, then $\delta_{\Gamma}(f_5) = 6$ because $\delta_{\Gamma}^r(f_5) = 5$ and $\delta_{\Gamma}^c(f_5) = 1$. (b) Drawing Γ_2 is obtained from Γ_1 by rerouting the edge *uv* of *G*. Γ_2 is a 1⁺-real face drawing and has $\nu = 20$ vertices (with 7 crossing-vertices), $\mu = 33$ edges, and $\varphi = 18$ faces.

A. DEGREE OF VERTICES AND FACES

For a vertex $v \in V(G)$, denote by $\delta_G(v)$ the degree of v in G, i.e., the number of edges incident to v. Analogously, for a vertex $v \in V(\Gamma)$, let $\delta_{\Gamma}(v)$ be the *degree of* v in Γ . For a face $f \in F(\Gamma)$, let $\delta_{\Gamma}(f)$ denote the *degree of f*, i.e., the number of times we traverse vertices (either real- or crossingvertices) while walking on the boundary of f clockwise. Each vertex contributes to $\delta_{\Gamma}(f)$ the number of times we traverse it (possibly more than once if the boundary of f is not a simple cycle); see face f_5 in Figure 2(a) for an example. Also, denote by $\delta_{\Gamma}^{r}(f)$ the real-vertex degree of f, i.e., the number of times we traverse a real-vertex of Γ while walking on the boundary of f clockwise. Again, each real-vertex contributes to $\delta_{\Gamma}^{r}(f)$ the number of times we traverse it. Finally, $\delta_{\Gamma}^{c}(f)$ denotes the number of times we traverse a crossing-vertex of Γ while walking on the boundary of f clockwise. Clearly, $\delta_{\Gamma}(f) = \delta_{\Gamma}^{r}(f) + \delta_{\Gamma}^{c}(f).$

B. K⁺-FACE DRAWINGS

Given a graph *G* and a positive integer *k*, a k^+ -real face drawing of *G* is a drawing Γ of *G* such that the boundary of each face of Γ has at least *k* real-vertices. If *G* admits a k^+ -real face drawing, we say that *G* is a k^+ -real face graph. An outer k^+ -real face drawing of *G* is a k^+ -real face drawing Γ of *G* such that all its real-vertices are on the boundary of the external face. If *G* admits an outer k^+ -real face drawing we say that *G* is an outer k^+ -real face drawing we say that *G* is an outer k^+ -real face drawing we say that *G* is an outer k^+ -real face graph. We say that a face $f \in F(\Gamma)$ is an *h*-real face, where *h* is a non-negative integer, if $\delta_{\Gamma}^r(f) = h$. An *h*-real face of degree *d* is called an *h*-real *d*-gon. An *h*-real 3-gon is also called an *h*-real triangle, and an *h*-real 4-gon is also called an *h*-real edge ($h \in \{0, 1, 2\}$) if $|\{u, v\} \cap V(G)| = h$, that is, *e* contains *h* real-vertices.

IV. DENSITY OF K⁺-REAL FACE GRAPHS

This sections gives tight upper bounds on the number of edges that a k^+ -real face graph can have. All our proofs exploit the well-known Euler's formula for planar drawings, which relates the number of edges, vertices, and faces. In our case, the idea is to use several applications of this formula on (nonplanar) k^+ -real face drawings, where the vertices are both the real-vertices and the crossing-vertices of the drawing, and where the edges of the drawing are portions of edges of the input graph.

We start by proving an upper bound on the number χ of crossing-vertices in a k^+ -real face drawing. It will be exploited to prove some of our edge density results.

Lemma 1: For any given k^+ -real face drawing drawing Γ of a graph *G*, the following inequality holds:

$$\chi \le \frac{2-k}{k} \cdot m + n - 2 \tag{1}$$

Proof: By hypothesis, each face $f \in F(\Gamma)$ contains at least k real-vertices. Since each real-vertex $v \in V(G)$ can belong to at most $\delta_G(v)$ faces of Γ and since $\sum_{v \in V(G)} \delta_G(v) = 2m$, we have that the number φ of faces of Γ is such that $\varphi \leq \frac{2m}{k}$. Also, the number of edges μ of Γ is such that $\mu = m + 2\chi$. Hence, by Euler's formula applied to Γ , we have $\varphi = \mu + 2 - v = m + 2\chi + 2 - n - \chi$, and hence $\varphi = m + \chi + 2 - n$. It follows that $\chi = \varphi - m + n - 2 \leq \frac{2m}{k} - m + n - 2 = \frac{2-k}{k} \cdot m + n - 2.\Box$

A. K^+ -REAL FACE GRAPHS, WITH $K \ge 2$

We first prove the edge density upper bound for the the case $k \ge 3$ (Theorem 1) and then we consider the case k = 2 (Theorem 2).

Theorem 1: Let k be an integer such that $k \ge 3$. If G is a k^+ -real face graph with n vertices and m edges, then $m \le \frac{k}{k-2}(n-2)$, and this bound is tight. Also, the k^+ -real face drawings of optimal *n*-vertex k^+ -real face graphs are exactly the *n*-vertex planar drawings in which each face is a simple k-gon.

Proof: Let Γ be any k^+ -real face drawing of G. When $k \ge 3$, the term $\frac{2-k}{k}$ is negative and, equivalently, $\frac{k}{k-2}$ is positive. Since the number χ of crossing-vertices of Γ cannot be negative, i.e., $\chi \ge 0$, by Equation (1) of Lemma 1 we have that the number m of edges of G must satisfy the inequality $m \le \frac{k}{k-2}(n-2)$.

To show that the bound is tight, consider any *n*-vertex planar graph with a given planar embedding, where each face has exactly k vertices. By Euler's formula, it is immediate to see that such a graph has $m = \frac{k}{k-2}(n-2)$ edges.

Finally, for an *n*-vertex graph with $m = \frac{k}{k-2}(n-2)$ edges, Equation (1) implies $\chi \le 0$. It follows that in every k^+ -real face drawing Γ of an optimal *n*-vertex k^+ -real face graph *G* the number χ of crossings equals 0, that is, *G* is necessarily planar. This implies that each face of Γ has degree *k*. Indeed, since each face of Γ has at least *k* vertices, if one of these faces had more than *k* vertices, then *G* would have more than $m = \frac{k}{k-2}(n-2)$ edges, which is not possible. \Box Theorem 2: If G is a 2^+ -real face graph with *n* vertices and *m* edges, then $m \le 4n - 8$, and this bound is tight. Also, the optimal *n*-vertex 2^+ -real face graphs are exactly the optimal 1-planar graphs.

Proof: Let Γ be any 2⁺-real face drawing of *G*. With k = 2, from Equation (1) of Lemma 1 we get $\chi \le n - 2$. Since $\mu \le 3\nu - 6$, since $\mu = m + 2\chi$, and since $\nu = n + \chi$, we have $m \le \chi + 3n - 6$, and therefore $m \le n - 2 + 3n - 6 = 4n - 8$. This proves the claimed upper bound on *m*.

To see that this bound is tight, consider the family of 1-planar graphs, which admit a drawing Γ with at most one crossing per edge. Each face of Γ contains at least two real-vertices (see also [75]), thus Γ is a 2⁺-real face drawing. In particular, for n = 8 and for every $n \ge 12$, there exists an optimal 1-planar graph with *n* vertices and 4n-8 edges [20]. To complete the proof, we show that every 2⁺-real face drawing Γ of an optimal 2⁺-real face graph G is also a 1-planar drawing of G, and hence that G is an optimal 1-planar graph. Suppose by contradiction that this is not true. Since G does not have multiple edges, this implies that there is at least one edge crossed twice in Γ , and hence a face of degree at least four in Γ (namely a face with at least two real-vertices and two crossing-vertices). Since each other face of Γ has at least degree three and since the sum of the degrees of the faces of Γ equals twice the number μ of its edges, we have $2\mu \ge 3(\varphi - 1) + 4 = 3\varphi + 1$. Since $\mu = m + 2\chi$ and $\varphi = m + \chi + 2 - n$, we get $m \le \chi - 7 + 3n$. From the optimality of Γ , we can plug m = 4n - 8 into the previous inequality, and we get $\chi \ge n-1$, which contradicts Equation (1) of Lemma 1 when k = 2. \square

B. 1⁺-REAL FACE GRAPHS

To prove an upper bound on the number of edges in 1⁺real face graphs, we exploit a *discharging* technique, which has been successfully used in previous papers on other beyond-planar graph families. See for example [22], [67], and [44]. Intuitively, a discharging technique consists of assigning an initial charge to the faces of the drawing, mainly based on the number of real- and crossing-vertices of their boundaries, and then moving some charges from faces with high capacity to faces with smaller capacities, without increasing the total charge. This idea, combined with Euler's formula, is used to count the maximum number of edges in the graph. More formally, as in [22], we consider a *charging function* ch : $F(\Gamma) \rightarrow \mathbb{R}$ such that, for each $f \in F(\Gamma)$, we set:

$$\operatorname{ch}(f) = \delta_{\Gamma}(f) + \delta_{\Gamma}^{r}(f) - 4 = 2\delta_{\Gamma}^{r}(f) + \delta_{\Gamma}^{c}(f) - 4 \qquad (2)$$

The value ch(f) is called the *initial charge* of f. In practice, each crossing-vertex of f contributes exactly once to the initial charge of f, while each real-vertex contributes twice. By using Euler's formula, it can be easily seen that the following relationship holds (refer to [22] for details):

$$\sum_{f \in F(\Gamma)} \operatorname{ch}(f) = 4n - 8 \tag{3}$$



FIGURE 3. (a) A 1⁺-real face drawing Γ . (b) An example of augmentation of Γ as described in the proof of Theorem 3; the added edges are gray and thick.

The aim of a discharging technique is to derive from the initial charging function ch a new function ch' that satisfies the next two properties (see also [22]):

- C1. ch'(f) ≥ αδ^r_Γ(f), for some real number α > 0;
 C2. Σ_{f∈F(Γ)} ch'(f) ≤ Σ_{f∈F(Γ)} ch(f)

If $\alpha > 0$ is a real number for which a charging function ch' satisfies C1 and C2, by Equation (3) we have: 4n - 8 = $\sum_{f \in F(\Gamma)} \operatorname{ch}(f) \geq \sum_{f \in F(\Gamma)} \operatorname{ch}'(f) \geq \alpha \sum_{f \in F(\Gamma)} \delta_{\Gamma}^{r}(f). \text{ Also,}$ since $\sum_{f \in F(\Gamma)} \delta_{\Gamma}^{r}(f) = \sum_{v \in V(G)} \delta_{G}(v) = 2m, \text{ we get the}$ following:

$$m \le \frac{2}{\alpha}(n-2) \tag{4}$$

Thus, Equation (4) can be exploited to prove upper bounds on the edge-density of a graph for specific values of α , whenever we find a charging function ch' that satisfies C1 and C2. We are now ready to present the main result of this section.

Theorem 3: If G is a 1^+ -real face graph with n vertices and *m* edges, then $m \le 5n - 10$, and this bound is tight.

Proof: Let Γ be a 1⁺-real face drawing of G. We first augment Γ and G with extra edges as follows (refer to Figure 3 for an example). If some face f of Γ contains a pair u and v of real-vertices but does not contain an edge uv on its boundary, then we augment Γ (and G) with an edge uv drawn in the interior of f, in such a way that it does not create any crossing. We then repeat this process until every pair of real-vertices in each face f is connected by an edge on the boundary of f. Note that this augmentation is not unique and may introduce multiple edges in G. However, it does not create any 0-real faces and any faces of degree two in the drawing; also, the drawing remains a 1^+ -real face drawing. Denoted by Γ' the drawing resulting from the edge augmentation on Γ , for each face $f \in F(\Gamma')$ we have that $\delta_{\Gamma'}(f) \geq 3$ and $1 \leq \delta_{\Gamma'}^r(f) \leq 3$. Also, denoted by G' the graph resulting from the augmentation on G, we have V(G') = V(G) and $E(G) \subseteq E(G')$; hence, an upper bound on the number of edges m' of G' is also an upper bound on the number of edges m of G.

Suppose given on Γ' the initial charging function ch : $F(\Gamma') \rightarrow \mathbb{R}$ of Equation (2). If we are able to define a charging function $ch' : F(\Gamma') \to \mathbb{R}$ that satisfies C1 and C2



FIGURE 4. Illustration for the proof of Theorem 3.

for $\alpha = \frac{2}{5}$, then by Equation (4) we get $m \le m' \le 5n - 10$, and we are done. We show how to define ch'.

For every face $f \in \Gamma'$, we initially set ch'(f) = ch(f) = $2\delta_{\Gamma'}^r(f) + \delta_{\Gamma'}^c(f) - 4$. With this choice and with $\alpha = \frac{2}{5}$, function ch' satisfies C2. Also, C1 becomes $2\delta_{\Gamma'}^r(f) + \delta_{\Gamma'}^c 4 \geq \frac{2}{5}\delta_{\Gamma'}^r(f)$, that is, $8\delta_{\Gamma'}^r(f) + 5\delta_{\Gamma'}^c(f) \geq 20$. Hence, since $\delta_{\Gamma'}(f) \geq 3$, C1 is always satisfied for each face f such that either $\delta_{\Gamma'}^r(f) \ge 2$, or $\delta_{\Gamma'}^r(f) = 1$ and $\delta_{\Gamma'}^c(f) \ge 3$. It follows that, the only faces that do not satisfy C1 are the 1-real triangles, i.e., each face t for which $\delta_{\Gamma'}^r(t) = 1$ and $\delta_{\Gamma'}^c(t) = 2$. Indeed, for a 1-real triangle t the initial charge equals 0, thus we need to suitably increase the value of ch'(t).

For each 1-real triangle t, let f be the face incident to the unique 0-real edge of t; see Figure 4(a). Note that it must be $\delta_{\Gamma'}(f) \geq 4$. Indeed, if it were $\delta_{\Gamma'}(f) = 3$ then G would contain two multiple edges (which is impossible because G is simple) or there would be two adjacent edges of G that cross in Γ (which is impossible because Γ is a simple drawing). Also, since Γ' is a 1⁺-real face drawing, we have $\delta_{\Gamma'}^r(f) \ge 1$. We apply a discharging operation, by moving a fraction $\frac{2}{5}$ of charge from f to t across their shared 0-real edge. In this way, we set $ch'(t) = \frac{2}{5}$ and reduce ch'(f) by $\frac{2}{5}$. The total charge of Γ' determined by ch' does not change (hence **C2** is still satisfied) but now ch'(t) satisfies C1.

Since for a face f the reduction of ch'(f) by $\frac{2}{5}$ occurs across a 0-real edge of f, the number of times this happens is at most $\delta_{\Gamma'}^{c}(f) - 1$. Therefore, after we have applied a discharging operation for each 1-real triangle, the charge ch'(f) of each face f of degree at least four is such that:

$$ch'(f) \ge 2\delta_{\Gamma'}^{r}(f) + \delta_{\Gamma'}^{c}(f) - 4 - \frac{2}{5}\delta_{\Gamma'}^{c}(f) + \frac{2}{5} = 2\delta_{\Gamma'}^{r}(f) + \frac{3}{5}\delta_{\Gamma'}^{c}(f) - \frac{18}{5}$$

Hence f satisfies C1 (i.e., $ch'(f) \ge \frac{2}{5}\delta_{\Gamma'}^r(f)$) if this relation holds:

$$8\delta_{\Gamma'}^r(f) + 3\delta_{\Gamma'}^c(f) \ge 18\tag{5}$$

It can be easily verified that the above relation is always satisfied for a face f of degree at least four, except when fis a 1-real quadrilateral (which consists of one real-vertex and 3 crossing-vertices). Indeed, if f is a 1-real quadrilateral it could have moved a fraction $\frac{2}{5}$ of charge towards a 1-real triangle t_1 and a fraction $\frac{2}{5}$ of charge towards another 1-real triangle t_2 ; see Figure 4(b). Both t_1 and t_2 share a crossingvertex x with f and with another face f'. In this case ch'(f) =



FIGURE 5. (a) A pentagon of an optimal 2-planar drawing with five inner chords that create a 0-real face. (b) To transform the 2-planar drawing into a 1^+ -real face drawing, it is sufficient to add a vertex (in gray) inside the 0-real face of each pentagon an connect this vertex to the real-vertices of the pentagon.

ch(f) $-\frac{4}{5} = 1 - \frac{4}{5} = \frac{1}{5} = \frac{2}{5} \delta_{\Gamma'}^r(f) - \frac{1}{5}$, thus f has a deficit of $\frac{1}{5}$. Observe that the boundary of f' contains two real-vertices adjacent to x, which are connected by an edge due to the edge augmentation initially performed on Γ . Hence, f' is a 2-real triangle and at this point we have ch'(f') = ch(f') = $1 = \frac{2}{5} \delta_{\Gamma'}^r(f') + \frac{1}{5}$. It follows that ch'(f') has a surplus of $\frac{1}{5}$, and we can move this surplus from f' to f, i.e., we increase ch'(f) by $\frac{1}{5}$ and decrease ch'(f') by $\frac{1}{5}$. Since this reduction of ch(f') can happen at most once for f', both f and f' satisfy **C1** at the end of this operation. This completes the proof that $m \le 5n - 10$.

To show the tightness of the bound, consider the class of *n*-vertex optimal 2-planar graphs, that is, graphs that admit a drawing with at most two crossings per edge and that have the maximum possible number of edges over all 2-planar graphs with *n* vertices. Pach and Tóth [20] show that these graphs have 5n - 10 edges. Also, Bekos et al. [76] show that an optimal 2-planar drawing Γ has a regular structure, namely: The crossing-free edges of Γ form a pentangulation P(G) (i.e., every face is a simple cycle of degree five), and inside each face of P(G) there are five crossing edges; see Figure 5(a). We can transform an optimal 2-planar drawing Γ into a 1⁺real face drawing Γ' by inserting, for each pentagon p of P(G), a real vertex inside the 0-real face of Γ that is contained in the interior of p, and by connecting this vertex to each of the five vertices of p; see Figure 5(b). If P(G) consists of h faces (pentagons), the augmented drawing has n' = n + h vertices and m' = 5n' - 10 edges.

V. DENSITY OF OUTER K⁺-REAL FACE GRAPHS

In this section we provide tight upper bounds on the maximum number of edges that an outer k^+ -real face graph can have, depending on k. For an outer k^+ -real face drawing Γ of a graph G, we denote by $F_{int}(\Gamma) \subset F(\Gamma)$ the subset of internal faces of Γ . Additionally, φ_{int} denotes the number of internal faces of Γ , that is $\varphi_{int} = |F_{int}(\Gamma)|$. Note that $\varphi = \varphi_{int} + 1$. As for k^+ -real face graphs, we first give an upper bound on the number χ of crossing-vertices in an outer k^+ -real face drawing, which will be exploited to prove some edge density results in this constrained scenario.

Lemma 2: Let G be a graph and let k be any positive integer. If Γ is an outer k^+ -real face drawing of G then the

following holds:

$$\chi \le \frac{2-k}{k} \cdot m + \frac{k-1}{k} \cdot n - 1 \tag{6}$$

Proof: By hypothesis, each face $f \in F_{int}(\Gamma)$ contains at least k real-vertices. Since each real-vertex $v \in V(G)$ can belong to at most $\delta_G(v) - 1$ internal faces of Γ and since $\sum_{v \in V(G)} (\delta_G(v) - 1) = 2m - n$, we have that the number φ_{int} of internal faces of Γ is such that $\varphi_{int} \leq \frac{2m - n}{k}$, and therefore $\varphi \leq \frac{2m - n}{k} + 1$. By Euler's formula applied to Γ , we have $\varphi = \mu + 2 - v = m + 2\chi + 2 - n - \chi$, and hence $\varphi = m + \chi + 2 - n$. It follows that $\chi = \varphi - m + n - 2 \leq \frac{2m - n}{k} - m + n - 1 = (1 - \frac{1}{k}) \cdot n - (1 - \frac{2}{k}) \cdot m - 1 = \frac{2-k}{k} \cdot m + \frac{k - 1}{k} \cdot n - 1$.

A. OUTER K^+ -REAL FACE GRAPHS, WITH $K \ge 2$

As for the unconstrained scenario, we first consider the case $k \ge 3$ and then the case k = 2.

Theorem 4: Let k be a positive integer such that $k \ge 3$. If G is an outer k^+ -real face graph with n vertices and m edges, then $m \le \frac{k-1}{k-2} \cdot n - \frac{k}{k-2}$, and this bound is tight. Also, the optimal n-vertex outer k^+ -real face drawings are exactly the n-vertex outerplanar drawings whose internal faces are simple k-gons.

Proof: Let Γ be any outer k^+ -real face drawing of G, with $k \ge 3$. Since the number χ of crossing-vertices of Γ cannot be negative, by Equation (6) of Lemma 2 we have that $\frac{2-k}{k} \cdot m + \frac{k-1}{k} \cdot n - 1 \ge 0$. Since $\frac{2-k}{k}$ is negative, this implies that $m \le (\frac{k-1}{k} \cdot n - 1)\frac{k}{k-2}$, and therefore $m \le \frac{k-1}{k-2} \cdot n - \frac{k}{k-2}$. To see that the bound is tight, consider the family of *n*-

To see that the bound is tight, consider the family of *n*-vertex outerplane graphs (i.e., outerplanar embedded graphs) where each internal face has exactly *k* vertices. For such an *n*-vertex graph, we have $2m = k\varphi_{int} + n$. Also, $\varphi_{int} + 1 = m + 2 - n$, and hence $\varphi_{int} = m - n + 1$. Thus, 2m = k(m - n + 1) + n, which yields $m = \frac{k-1}{k-2} \cdot n - \frac{k}{k-2}$. The final part of the theorem follows from the same considerations as in the proof of Theorem 1.

Theorem 5: Let G be an outer 2^+ -real face graph with n vertices and m edges. We have that $m \le 2.5n - 4$, and this bound is tight. Also, the n-vertex optimal outer 2^+ -real face graphs are exactly the optimal outer-1-planar graphs.

Proof: Let Γ be any outer 2⁺-real face drawing of G. By Lemma 2, with k = 2, we get $\chi \leq \frac{n}{2} - 1$. If we remove from Γ exactly one edge of G per crossing-vertex, we get an outerplanar graph with $m' = m - \chi$ edges and n vertices. Since a maximal outerplanar graph with n vertices has at most 2n - 3 edges, we have $m - \chi \leq 2n - 3$, and therefore $m \leq 2n - 3 + \chi \leq 2n - 3 + \frac{n}{2} - 1$, that is, $m \leq \frac{5}{2}n - 4 = 2.5n - 4$. This proves that 2.5n - 4 is an upper bound on the number of edges of G.

To show that this bound is tight, we describe an infinite family of outer 2^+ -real face graphs whose number of edges matches the bound. Refer to Figures 6(a) and 6(b). Let $h \ge 1$ be any positive integer. Construct a graph G_h with n = 2h + 2 vertices and m = 2.5n - 4 edges as follows. Suppose that in a drawing Γ of G_h the *n* vertices are distributed along a circumference such that *u* and *v* are



FIGURE 6. An optimal outer 2^+ -real face graph with (a) n = 8 vertices, and (b) n = 12 vertices. An optimal outer 1^+ -real face graph with (c) n = 8 vertices, and (d) n = 11 vertices.

the bottom-most and the top-most vertices, respectively; this means that there is a set V_l of h vertices encountered walking from u to v on the circumference clockwise, and another set V_r of h vertices encountered walking from u to v on the circumference counterclockwise. The m edges of G_h are as follows. There is a cycle of n edges that contains all n vertices of G_h ; it forms the boundary of the external face of Γ . There are other n - 3 internal edges that connect u to all vertices of G_h that are not adjacent to u on the external boundary. Finally, there are h edges, each connecting a pair of distinct vertices in $V_l \cup V_r$ that have a topological distance equal to two on the external cycle, but not the two vertices adjacent to u. Hence, m = n + (n - 3) + h = 2.5n - 4.

We finally prove that every *n*-vertex optimal outer 2^+ -real face drawing Γ is also 1-planar; since optimal outer-1-planar graphs have at most 2.5n - 4 edges [63], [77], this will imply that optimal outer 2^+ -real face graphs are exactly the optimal outer-1-planar graphs. Suppose, by contradiction, that an edge of *G* crosses twice in Γ . This implies that there is an internal face of Γ of degree at least 4. Therefore $2\mu \ge 3(\varphi - 2) + 4 + n$, which leads to $2\mu \ge 3\varphi - 2 + n$. Since $\mu = m + 2\chi$ and $\varphi = m + \chi + 2 - n$, we have $\chi \ge m + 4 - 2n$. Since Γ is optimal, then m = 2.5n - 4, and hence $\chi \ge 0.5n$, which contradicts Lemma 2 for k = 2.

B. OUTER 1⁺-REAL FACE GRAPHS

As for 1⁺-real face graphs, we use a discharging technique to prove an upper bound on the number of edges of outer 1⁺-real face graphs. An outer 1⁺-real face Γ is *edge-maximal* if the drawing obtained by adding to Γ any new edge between two of its real-vertices is no longer outer 1⁺-real face. An example of edge-maximal outer 1⁺-real face drawing is illustrated in Figure 7(a). However, this graph is not optimal, because for any $n \geq 3$ there exist outer 1⁺-real face graphs that contain 3n - 6 edges (Theorem 6).

The next lemma is fundamental to prove Theorem 6. It shows some important structural properties of edgemaximal outer 1^+ -real face drawings. For an illustration of the properties stated in the lemma refer to Theorem 7.

Lemma 3: Let *G* be an *n*-vertex outer 1^+ -real face graph, with $n \ge 4$, and let Γ be an edge-maximal outer 1^+ -real face drawing of *G*. The following properties hold:

- a) The boundary of the external face is a simple cycle without crossing vertices and with exactly *n* real-vertices.
- **b)** Each internal face of Γ is either a 3-real triangle, or a 2-real *d*-gon ($d \ge 3$), or a 1-real triangle, or a 1-real quadrilateral.
- c) We can map each 1-real triangle to exactly one face of Γ that is either a 2-real *d*-gon, for $d \ge 4$, or a 1-real quadrilateral, in such a way that: (*i*) at most (*d*-3) 1-real triangles are mapped to the same 2-real *d*-gon; and (*ii*) at most two 1-real triangles are mapped to the same 1-real quadrilateral.
- d) The number of 3-real triangles plus the number of 2-real *d*-gons is exactly *n*, and the number of 1-real quadrilaterals is at most n 4.

Proof: Let f_0 be the external face of Γ . Suppose by contradiction that the boundary of f_0 contains a crossingvertex c. Let u be the first real-vertex encountered starting from c and moving on the boundary of f_0 clockwise. Analogously, let v be the first real-vertex encountered starting from c and moving on the boundary of f_0 counterclockwise. Note that Γ does not have an edge *uv*. Indeed, since we are assuming that c is on the external face, if Γ contained uv, then either Γ would not be simple (which is impossible) or some real-vertices would not be on the external face (which contradicts the fact that Γ is outer 1⁺-real face). At this point we can easily augment Γ by adding edge uv in the external face while keeping the drawing outer 1⁺-real face, which contradicts the fact that Γ is edge-maximal. This proves that the boundary of f_0 consists of exactly *n* realvertices and no crossing-vertices. Also, if the boundary of f_0 were not a simple cycle, we could add a minimal subset of edges in the external face to achieve this property, while keeping all real-vertices on the external boundary. Again, this contradicts the edge-maximality of Γ and completes the proof of Property (a).

We now turn our attention to Property (b). Let f be any internal face of Γ . We first observe that either f has at most two real-vertices or f is a 3-real triangle. Indeed, suppose that f contains (at least) three real-vertices. If f is not a triangular face than there must be two real-vertices u and v in f that are not connected by an edge. Hence, we can augment Γ by adding an edge uv that splits f; the new drawing



FIGURE 7. (a) An edge-maximal outer 1⁺-real face drawing of a graph with 8 vertices and 17 edges. The faces filled in gray represent 1-real triangles. Each face f_i with $i \in \{1, ..., 7\}$ is a 2-real triangle. The faces f_8 , f_9 , f_{10} are 1-real quadrilateral, and f_{11} is a 2-real 4-gon. (b) A mapping of the 1-real triangles to the other faces of the drawing; the mapping, showed by blue arrows, satisfies property (c) of Lemma 3. (c) Another example of an edge-maximal outer 1⁺-real face drawing, with a mapping that satisfies property (c) of Lemma 3 is shown.

would still be an outer 1⁺-real face drawing of *G*, which contradicts the edge-maximality of Γ . Therefore, if *f* contains three real-vertices then it must be a 3-real triangle. Suppose now that *f* has exactly two real-vertices *u* and *v*. In this case, by Property (a), *f* has an edge *uv* shared with the external face; also, since *f* has degree $d \ge 3$, the remaining d - 2 vertices of *f* are all crossing-vertices, i.e., *f* is a 2-real *d*-gon. Finally consider a face *f* with exactly one real-vertex *v* and suppose for contradiction that it is neither a 1-real triangle nor a 1-real quadrilateral. This implies that *f* has degree at least five, and that there is a 0-real edge *e* in Γ shared by *f* and another face *f'* that contains a real vertex *u* not adjacent to *v*. If we augment Γ with an edge *uv* that crosses *e* we get a new outer 1⁺-real face drawing with one edge more than Γ , a contradiction. This completes the proof of Property (b).

We now focus on Property (c). Let *t* be any 1-real triangle of Γ , let *e* be the 0-real edge of *t*, and let *f* be the face of Γ sharing *e* with *t*. Since *G* has no multiple edges, *f* cannot be a 1-real triangle. Also, since the boundary of *f* contains at least two crossing-vertices, by Property (b), *f* is neither a 3-real triangle nor a 2-real 3-gon. Hence, by Property (b), *f* is either a 2-real *d*-gon, with $d \ge 4$, or a 1-real quadrilateral. We map *t* on *f*. In this way, every 1-real triangle is either mapped to exactly one 2-real *d*-gon, with $d \ge 4$, or to exactly 1-real quadrilateral. Moreover, since each 2-real *d*-gon *f*, with $d \ge$ 4, has exactly (*d*-3) 0-real edges on its boundary, at most (*d*-3) 1-real triangles are mapped to *f*. Analogously, since each 1-real quadrilateral *f* has exactly two 0-real edges, at most two 1-real triangles are mapped to *f*. This completes the proof of Property (c).

It remains to prove Property (d). By Property (a), the boundary of the external face f_0 consists of exactly n 2-real edges. By Property (b), any 2-real edge is incident to an internal face that is either a 3-real triangle or a 2-real d-gon, with $d \ge 3$. Also, since all real-vertices of Γ belong to f_0 , each 3-real triangle is adjacent to f_0 . If there is no 3-real triangle in Γ , then each edge of f_0 is incident to a 2-real d-gon and, since no two 2-real d-gons share a 2-real edge, there are exactly n 2-real d-gons. Suppose vice versa that there are some 3-real triangles. Each 3-real triangle is incident to two edges of f_0 and to a third 2-real edge shared with a 2-real d-gon. Hence, for each pair of external edges of a 3-real triangle f, we count a pair of internal faces of Γ consisting of f and of the 2-real d-gon adjacent to f. This implies that the number of 3-real triangles plus the number of 2-real d-gons of Γ equals n. Finally, consider the number of 1-real quadrilaterals in an outer 1⁺-real face drawing. To form a single 1-real quadrilateral we need at least 5 vertices, and it can be seen that any additional 1-real quadrilateral requires to use at least one more vertex. Hence a drawing with a number q of 1-real quadrilaterals requires at least n = q + 4 vertices, that is, $q \le n - 4$.

Theorem 6: If G is an outer 1⁺-real face graph with n vertices and m edges, then $m \leq 3n - 6$, and this bound is tight.

Proof: To prove the upper bound on the maximum number of edges, it is enough to concentrate on edge-maximal outer 1⁺-real face drawings of *G*. Let Γ be such a drawing. If *G* has three vertices only, then Γ is a planar 3-cycle and the statement trivially holds. Assume that $n \ge 4$. We exploit a discharging technique like in the proof of Theorem 3. However, in this case, we aim to show the existence of a charging function ch' that satisfies **C1** and **C2** for $\alpha = \frac{2}{3}$. If such a function exists then, by Equation (4), we get $m \le 3n - 6$. For each face $f \in F(\Gamma)$, initially set ch'(f) = ch(f), where ch(f) is the charging function of Equation (2). Denote by f_0 the external face of Γ . Based on Properties (a) and (b) of Lemma 3, it holds $\delta_{\Gamma}(f_0) = \delta_{\Gamma}^{r}(f_0) = n$ and each internal face of Γ is either a 3-real triangle, or a 2-real d-gon, or a 1-real triangle, or a 1-real quadrilateral. At this point we have:

- $\operatorname{ch}'(f_0) = 2\delta_{\Gamma}^r(f_0) + \delta_{\Gamma}^c(f_0) 4 = 2n 4$; the charge excess of f_0 with respect to $\frac{2}{3}\delta_{\Gamma}^r(f_0)$ is $\frac{4}{3}n 4$.
- If f is a 3-real triangle, then ch'(f) = 2; the charge excess (and the charge deficit) of f is zero.
- If f is a 2-real d-gon, then ch'(f) = d-2; hence, if d = 3 (i.e., f is a 2-real triangle) then f has a charge deficit of $\frac{1}{3}$, while if $d \ge 4$ then it has an excess of $d \frac{10}{3}$.
- If f is a 1-real triangle then ch(f) = 0 and f has a charge deficit of $\frac{2}{3}$.
- If f is a 1-real quadrilateral then ch(f) = 1 and f has a charge excess of ¹/₂.

We now want to modify ch' by moving charges from faces with an excess to faces with a deficit, in such a way that C1 is satisfied. Based on the aforementioned analysis, the only faces with a deficit are the 2-real triangles (with a deficit of $\frac{1}{3}$) and the 1-real triangles (with a deficit of $\frac{2}{3}$).

Consider a mapping of each 1-real triangle to either a 2-real d-gon (with $d \ge 4$) or to a 1-real quadrilateral that satisfies the conditions (*i*) and (*ii*) in Property (c) of Lemma 3. We do the following:

- For every 1-real triangle t mapped to a 2-real d-gon f, we decrease ch'(f) by $\frac{2}{3}$ and increase ch'(t) by $\frac{2}{3}$. After this operation, ch'(t) satisfies C1. Also, the charge ch'(f) is decreased by at most $(d - 3)\frac{2}{3}$. Since the excess of f before the operation was equal to $d - \frac{10}{3}$, the new excess after the operation is at least $d - \frac{10}{3} - (d - 3)\frac{2}{3} = \frac{1}{3}d - \frac{4}{3}$, which is always non-negative for $d \ge 4$. Hence, ch'(f) still satisfies C1.
- For every 2-real triangle *t*, we decrease $ch'(f_0)$ by $\frac{1}{3}$ and increase ch'(t) by the same quantity. After this operation, ch'(t) satisfies **C1**. Also, since by Property (d) of Lemma 3. the number of 2-real triangles is at most *n*, the excess of f_0 after this operation becomes n 4. For each 1-real quadrilateral *f*, we further move one unit of charge from f_0 to *f*. After this operation, the excess of each 1-real quadrilateral grows from $\frac{1}{3}$ to $\frac{4}{3}$. At the same time, since by Property (d) of Lemma 3. there are at most (n 4) 1-real quadrilaterals, the excess of f_0 either remains positive or becomes 0; hence, $ch'(f_0)$ still satisfies **C1**.
- Finally, for 1-real triangle t mapped to a 1-real quadrilateral f, we increase ch'(t) by $\frac{2}{3}$, so that it now satisfies C1, and we decrease ch'(f) by $\frac{2}{3}$. Since there are at most two 1-real triangles mapped to f, the charge of f is decreased by at most $\frac{4}{3}$, and therefore the excess of f either remains positive or it becomes zero, which still satisfies C1.

This completes the proof that $m \leq 3n - 6$. To prove that the bound is tight, for each $n \ge 3$ we describe how to construct an outer 1^+ -real face graph with *n* vertices and m = 3n - 6 edges. If n = 3 then G is just a triangle, while if n = 4 then G is K_4 (which can be drawn with all vertices on the boundary of the external face and a single edge crossing inside). Let $n \ge 5$. The construction of G is as follows; see Figures 6(c) and 6(d). Start with a cycle of *n* vertices, which defines the boundary of the external face in an outer 1⁺-real face drawing Γ of G. Arbitrarily select a vertex v of G and augment Γ (and G) with n-3 internal edges that connect v to all vertices that were not already adjacent to it. Finally, let *u* be the vertex that immediately follows *v* while walking on the external boundary of Γ clockwise, and let w be the vertex that immediately precedes v by walking on the external boundary of Γ counterclockwise. Augment Γ (and G) with two interlaced chains of edges, each connecting a sequence of pairs of vertices that have topological distance equal to two on the external face but excluding the pair $\{u, w\}$ and any other pair that involves v (which is already connected to any other vertex). If n is odd, each of these two chains consists of $\frac{n-3}{2}$

edges. If *n* is even, one of the chain consists of $\frac{n-2}{2}$ edges and the other consists of $\frac{n-2}{2} - 1$ edges. It is immediate to check that, with this construction, *G* has 3n - 6 edges and Γ is an outer 1⁺-real face drawing.

VI. COMPLETE GRAPHS AND COMPLETE BIPARTITE GRAPHS

In this section we study complete graphs and complete bipartite graphs. For the complete graphs we provide a characterization of those that admit k^+ -real face drawings, for any $k \ge 1$. Concerning complete bipartite graphs, for most of them we are able to establish whether they belong or not to the different k^+ -real face graph families; the question remains open for few instances.

A. COMPLETE GRAPHS

Denote by K_n the complete graph on *n* vertices. The number of edges of K_n is $m = \frac{n(n-1)}{2}$. From the edge density results summarized in Table 1, the following facts immediately hold:

- K_n with $n \ge 9$ is not a k^+ -real face graph, for any $k \ge 1$.
- K_n with $n \ge 7$ is not a k^+ -real face graph, for any $k \ge 2$.
- K_n with $n \ge 5$ is not an outer k^+ -real face graph, for any $k \ge 1$.
- K_n with $n \ge 4$ is not an outer k^+ -real face graphs, for any $k \ge 3$.

On the positive side, we have the following:

- K_3 and K_4 are outer 2^+ -real face graphs (trivial).
- K_5 is a 2⁺-real face graph (see Figure 1).
- K_6 is a 2⁺-real face graph (see Figure 8(a)).
- K_7 is a 1⁺-real face graph (see Figure 8(b)).
- K_8 is a 1⁺-real face graph (see Figure 8(c)).

It might be interesting to observe that the k^+ -real face drawings shown in Figure 8(a), Figure 8(b), and Figure 8(c) correspond to drawings that match the *crossing number* for the graphs K_6 , K_7 , and K_8 , respectively. Recall that the crossing number of a graph G is the minimum number of crossings required by any drawing of G in the plane.

The discussion above implies the following characterization.

Theorem 7: The complete graphs that admit a k^+ -real face drawing or an outer k^+ -real face drawing (with $k \ge 1$) are those reported in Table 4.

The following corollary immediately holds from Theorem 7.

Corollary 1: For any $k \ge 1$ and $n \ge 3$, there exists an O(1)-time algorithm that decides whether the complete graph K_n is k^+ -real face or outer k^+ -real face.

B. COMPLETE BIPARTITE GRAPHS

Denote by $K_{a,b}$ the complete bipartite graph on n = a + b vertices, where *a* is the cardinality of one partition set and *b* is the cardinality of the other partition set. Recall that $K_{a,b}$ has $m = a \cdot b$ edges. Without loss of generality, in the following we always assume that $b \ge a$. We start with the following.



FIGURE 8. (a) 2⁺-real face drawing of K_6 ; (b) 1⁺-real face drawing of K_7 ; (c) 1⁺-real face drawing of K_8 .

TABLE 4. Complete graphs that admit a k^+ -real face drawing or an outer k^+ -real face drawing.

Graph	1 ⁺ -real face	2 ⁺ -real face	$k^{+}\text{-real}$ face ($k \ge 3$)	outer 1 ⁺ -real face	outer 2 ⁺ -real face
K_3	yes	yes	yes	yes	yes
K_4	yes	yes	yes	yes	yes
K_5	yes	yes	no	no	no
K_6	yes	yes	no	no	no
K_7	yes	no	no	no	no
K_8	yes	no	no	no	no
$K_{\geq 9}$	no	no	no	no	no

Lemma 4: For any $b \ge 2$, the graph $K_{2,b}$ is a k^+ -real face graph for $k \le 4$, whereas it is not k^+ -real face for $k \ge 5$.

Proof: The graph $K_{2,b}$ is a planar graph for every $b \ge 2$. Also, in every planar embedding of $K_{2,b}$ all faces have degree four. By Theorem 1, it follows that $K_{2,b}$ is a k^+ -real face graph for $k \le 4$, whereas it is not k^+ -real face for $k \ge 5$.

To study what happens for the complete bipartite graphs $K_{a,b}$ that are not planar, that is, those for which $a \ge 3$ and $b \ge a$, we exploit the upper bounds on the number of crossings summarized in Table 1, rather than their edge density. Namely, the famous Zarankiewicz's conjecture [78] claims that the crossing number of $K_{a,b}$ equals the value of the following function:

$$Z(a,b) = \left\lfloor \frac{a}{2} \right\rfloor \left\lfloor \frac{a-1}{2} \right\rfloor \left\lfloor \frac{b}{2} \right\rfloor \left\lfloor \frac{b-1}{2} \right\rfloor$$

Kleitman [79] showed that the conjecture holds for all values $b \leq 6$ and also proved that the smallest counterexample to Zarankiewicz's conjecture (if any) must occur for odd values of *a* and *b*. Furthermore, Woodall [80] used a computer program to verify the correctness of the conjecture for $K_{7,7}$ and $K_{7,9}$. Thus, the smallest unsettled instances of Zarankiewicz's conjecture are $K_{7,11}$ and $K_{9,9}$. Observe that the number of edges of $K_{9,9}$ is higher than the maximum number of edges of any 1⁺-real face graph, thus this graph (and every other complete bipartite graph with higher edge density) does not have a k^+ -real face drawing, for any $k \geq 1$. About $K_{7,11}$, we cannot use the same argument. However, by Lemma 1 we know that a 1⁺-real face drawing of a graph with *n* vertices and *m* edges has at most m + n - 2 crossings; hence any 1⁺-real face drawing of $K_{7,11}$ cannot have more than 93 crossings. Nonetheless, the crossing number of $K_{7,10}$ equals Z(7, 10) = 225, and the crossing number of $K_{7,11}$ is larger than the one of $K_{7,10}$. It follows that also $K_{7,11}$ is not 1⁺-real face, for any $k \ge 1$. Therefore, we can conclude that for every complete bipartite graph $K_{a,b}$ that might have a chance of being 1⁺-real face, we can assume that the crossing number of $K_{a,b}$ equals Z(a, b). We now prove the following.

Lemma 5: For $b \ge 3$ and $k \ge 3$, the graph $K_{3,b}$ is not k^+ -real face.

Proof: It is enough to show that the statement holds for k = 3. By Lemma 1, a 3⁺-real face drawing has at most $-\frac{m}{3} + n - 2$ crossings. Hence, for $b \ge 3$, the number of crossings allowed in a 3⁺-real face drawing is at most 1. However, the crossing number of $K_{3,b}$ is Z(3, b), which is always greater than 1 when $b \ge 4$. Hence $K_{3,b}$ is not 3⁺-real face if $b \ge 4$.

To complete the proof, we must show that $K_{3,3}$ is not 3^+ -real face. Suppose by contradiction that a 3^+ -real face drawing Γ of $K_{3,3}$ exists. As observed above, Γ has at most one crossing. Call *e* one of the two crossing edges of $K_{3,3}$ in Γ , and let Γ' be the drawing obtained from Γ by removing *e*. Γ' is planar and all its faces have degree four. Also, the two end-vertices of *e* do not belong to the same face of Γ' , as otherwise it would be possible to reinsert *e* in Γ' without crossings, which is impossible because $K_{3,3}$ is not planar. It follows that *e* cuts an edge of a degree-4 face in Γ' , by generating a triangular face in Γ with only two real-vertices and one crossing-vertex, a contradiction.

We also observe that for every graph $K_{a,b}$ with $a \ge 4$ (and $b \ge a$) the upper bound of Lemma 1 is strictly less than Z(a, b); thus $K_{a,b}$ is not a 3⁺-real face graph if $a \ge 4$. This observation, together with Lemma 4 and Lemma 5, immediately implies the following characterization.

Theorem 8: Let $k \ge 3$. The complete bipartite graph $K_{a,b}$, with $b \ge a$ is a k^+ -real face graph if and only if one of these two conditions holds: (*i*) $a \le 2$, $b \ge 2$, and k = 3; (*ii*) a = 2 and k = 4.

An implication of Theorem 8 is that there are no complete bipartite graphs that are k^+ -real face for $k \ge 5$. From Theorem 8 we get the following corollary.

Corollary 2: For any $k \ge 3$ and for any given complete bipartite graph $K_{a,b}$, there exists an O(1)-time algorithm that decides whether $K_{a,b}$ is k^+ -real face.

We now consider 2^+ -real face drawings of non-planar complete bipartite graphs. By Lemma 1, the number of crossings in any 2^+ -real face drawing of a graph with *n* vertices is at most n - 2. Still using Zarankiewicz's formula, it is immediate to verify that this upper bound is below the crossing number of $K_{a,b}$ when one of the following cases holds: (*i*) a = 3 and $b \ge 7$; (*ii*) a = 4 and $b \ge 5$; (*iii*) $a \ge 5$ and $b \ge a$;

On the other hand, each of the few remaining graphs, namely $K_{3,3}$, $K_{3,4}$, $K_{3,5}$, $K_{3,6}$, and $K_{4,4}$, admit a 2⁺-real face drawing, as shown in Figure 9. These considerations, together with Lemma 4, imply the following characterization of the complete bipartite graphs that are 2⁺-real face.

Theorem 9: A complete bipartite graph $K_{a,b}$, with $b \ge a$, is 2⁺-real face if and only if one of the following holds: (*i*) a = 2; (*ii*) a = 3 and $b \le 6$; (*iii*) a = 4 and b = 4.

The next corollary immediately follows.

Corollary 3: Given any complete bipartite graph $K_{a,b}$, there exists an O(1)-time algorithm that decides whether $K_{a,b}$ is 2^+ -real face.

We finally turn our attention to the study of the complete bipartite graphs that admit a 1⁺-real face drawing, in addition to those that also admit a k^+ -real face drawing for $k \in$ {2, 3, 4}. By Lemma 1, the number of crossings in any 1⁺real face drawing of a graph with *n* vertices and *m* edges is at most m + n - 2. As before, if we compare this upper bound with the crossing number given by Zarankiewicz's formula, we can conclude that the graph $K_{a,b}$ does not have a 1⁺-real face drawing in each of the following cases: (*i*) a = 3 and $b \ge 19$; (*ii*) a = 4 and $b \ge 13$; (*iii*) a = 5 and $b \ge 9$; (*iv*) a = 6 and $b \ge 7$; (*v*) $a \ge 8$ and $b \ge a$.

On the positive side, we are able to establish that each of the following complete bipartite graphs admits a 1⁺-real face drawing (see Figure 10 and Figure 11): $K_{3,7}$, $K_{3,8}$, $K_{3,9}$, $K_{3,10}$, $K_{4,5}$, $K_{4,6}$, $K_{4,7}$, $K_{5,5}$. It is interesting to observe that, as for the complete graphs, also all k^+ -real face drawings that we presented for the complete bipartite graphs match the crossing number, that is, they achieve the minimum number of crossings.

We leave it open to establish whether each of the following complete bipartite graphs is 1⁺-real face or not: (*i*) $K_{3,b}$ for $b \in [11, 18]$; (*ii*) $K_{4,b}$ for $b \in [8, 12]$, (*iii*) $K_{5,b}$ for $b \in [6, 8]$, and (*iv*) $K_{6,6}$.

We now consider the constrained scenario, that is outer k^+ -real face drawings of complete bipartite graphs. We characterize, for all values $k \ge 1$, the complete bipartite graphs that are outer k^+ -real face; for each graph listed below, we indicate the minimum k for which it is outer k^+ -real face.

- $K_{1,1}$ is outer 2⁺-real face (trivial).
- $K_{1,2}$ is outer 3⁺-real face (trivial).
- $K_{2,2}$ is outer 4⁺-real face (trivial).
- K_{2,3} and K_{2,4} are outer 2⁺-real face (see Figures 12(a) and 12(b)); it is easy to verify that any outer k⁺-real face

drawing of these two graphs has at least one face with only two real-vertices.

- $K_{2,5}$ and $K_{2,6}$ are outer 1⁺-real face (see Figures 12(c) and 12(d)); it is easy to verify that any outer k^+ -real face drawing of these two graphs has at least one face with one real-vertex.
- $K_{2,b}$ with $b \ge 7$ is not an outer k^+ -real face graph for any $k \ge 1$. It can be easily checked that in any drawing of this graph with all faces on the external face, there is at least one 0-real face.
- $K_{a,b}$ with $a, b \ge 3$ is not an outer k^+ -real face graph for any $k \ge 1$. It is sufficient to restrict to $K_{3,3}$ and verify that for any of the three non-symmetric circular orderings of black and white vertices along the boundary of the outer face, there is at least one 0-real face (see Figure 13).

Based on the above facts, the following immediately holds. *Corollary 4:* For any $k \ge 1$ and for any complete bipartite graph $K_{a,b}$, there exists an O(1)-time algorithm that decides whether $K_{a,b}$ is outer k^+ -real face.

VII. INCLUSION RELATIONSHIPS

We have observed that, by definition, the family of $(k + 1)^+$ real face graphs is properly included in the family of k^+ -real face graphs, for any integer $k \ge 1$. For k-planar graphs we have the opposite: each k-planar graph is also (k + 1)-planar. The results of Section IV yield some inclusion relationships between the families of k-planar graphs and k^+ -real face graphs, for $k \in \{1, 2\}$. Namely, Theorem 2 shows that 1planar graphs are 2^+ -real face graphs and that the sets of optimal 1-planar graphs and optimal 2^+ -real face graphs coincide. These relationships are summarized in Figure 14.

In the following, we prove a more general result about the relationship between k^+ -real face graphs and any other beyond-planar graph family with hereditary property. This result (Theorem 10) implies that, for any fixed positive integer k, there cannot exist a beyond-planar graph family with hereditary property that includes the whole set of k^+ real face graphs, independent of the maximum edge density of the two families. We now formalize this concept.

We say that a family \mathcal{F} of beyond-planar graphs has the *hereditary property* if any subgraph of a graph in \mathcal{F} also belongs to \mathcal{F} . Most of the beyond-planar graph families studied in the literature (see, e.g., [16]) have the hereditary property. Conversely, k^+ -real face graphs do not always satisfy this property; indeed, removing vertices from a k^+ -real face graph G may give rise to a subgraph G' whose drawings necessarily have a face with less than k vertices. However, it is immediate to see that any subgraph of G that has the same vertex set of G is still a k^+ -real face graph. We prove the following.

Lemma 6: For any integer k > 0 and for any family \mathcal{F} of beyond-planar graphs with hereditary property, there exists a k^+ -real face graph that does belong to \mathcal{F} .

Proof: Let G be any (connected) graph that does not belong to \mathcal{F} , i.e., $G \notin \mathcal{F}$. The idea is to augment G with

IEEEAccess



FIGURE 9. 2⁺-real face drawings of K_{3,3}, K_{3,4}, K_{3,5}, K_{3,6}, and K_{4,4}.



FIGURE 10. 1⁺-real face drawings for $K_{3,7}$, $K_{3,8}$, $K_{3,9}$, and $K_{3,10}$.





FIGURE 12. (a) and (b): outer 2^+ -real face drawings for $K_{2,3}$ and $K_{2,4}$, respectively. (c) and (d): outer 1^+ -real face drawings for $K_{2,5}$ and $K_{2,6}$, respectively.



FIGURE 13. The three possible (non-symmetric) outer k^+ -real face embeddings for $K_{3,3}$; 0-real faces are colored gray.



FIGURE 14. Venn diagram describing relationships between k^+ -real face graphs and k-planar graphs.

vertices and edges so that the resulting graph G' is a k^+ -real face graph. If we are able to do that, then G' cannot belong to \mathcal{F} , because otherwise also G would belong to \mathcal{F} (due to the hereditary property of \mathcal{F}), which contradicts our initial hypothesis.

To construct G' starts from any drawing Γ of G in the plane. If Γ is already a k^+ -real face drawing, we are done, i.e., G' coincides with G. Otherwise, we augment Γ (and hence G) into a new drawing Γ' as explained below; see Figure 15 for an example of this augmentation. We first consider the set of 0-real faces of Γ . If this set is not empty, there must be a 0-real face f that is adjacent to a face f' containing a real-vertex v. Add to Γ a new real-vertex u in the interior of f and connect u to v with an edge that crosses exactly one edge shared by f and f'. After this augmentation, the set of 0-real faces in the new drawing is decreased by one element. Repeat this procedure until there is no more 0-real faces in the drawing. Then, for every face f of Γ with h < k real-vertices (if any), arbitrarily select a real-vertex v of f, and attach to v a chain of k - h



FIGURE 15. Example of the augmentation procedure in the proof of Lemma 6. The example shows how to construct an instance in the family of 2^+ -real face graphs, which is not in the family of 1-planar graphs. (a) A drawing of a graph *G* that is not 1-planar; it has only one 0-real face (filled in gray). (b) An augmentation that removes the 0-real face; the new elements (a vertex and an edge) are in red. (c) A further augmentation that yields a 2^+ -real face drawing; for each face *f* with less than two vertices, we attach an extra real-vertex to an arbitrary real-vertex on the boundary of *f*.

vertices in the interior of f. This creates a new face f' in place of f, whose boundary has k real-vertices. Once all those faces have been processed, the underlying graph G' of the resulting drawing is a (connected) k^+ -real face graph.

Lemma 6 immediately implies the following.

Theorem 10: For any positive integer k, the family of k^+ -real face graphs is not included in any beyond-planar graph family with hereditary property.

We say that two graph families are *incomparable* if none of the two families includes the other. Theorem 10 allows us to easily prove a series of incomparability relationships.

Corollary 5: For any integers $k \ge 1$ and $h \ge 3$, the family of k^+ -real face graphs is incomparable with the families of *h*-planar graphs, min-*h*-planar graphs, and *h*-quasi planar graphs.

Proof: Theorem 10 proves the existence of 1^+ -real face graphs that are neither *h*-planar, nor min-*h*-planar, nor *h*-quasi planar. On the other hand, since the maximum number of edges of *n*-vertex *h*-planar graphs and min-*h*-planar graphs, for $h \ge 3$, can be greater than 5n - 10 [20], [27], [28], [66], there exist *h*-planar graphs and min-*h*-planar graphs that are not 1^+ -real face graphs (and hence that are not k^+ -real face graphs, for any $k \ge 2$). Similarly, *h*-quasi planar graphs, for any $h \ge 3$, can have higher density than 1^+ -real face graphs, because 3-quasi planar graphs can have up to 6.5n - 20 edges [21]. The claimed incomparabilities follow.

Corollary 6: For any $k \ge 2$, the family of k^+ -real face graphs is incomparable with each of the following families: (*i*) 2-planar graphs; (*ii*) min-2-planar graphs; (*iii*) fan-planar graphs; and (*iv*) *h*-gap planar graphs, for any $h \ge 1$.

Proof: As we have already seen, *n*-vertex 2-planar graphs can have up to 5n - 10 edges [52]. Similarly, min-2-planar graphs [28], fan-planar graphs [35], and 1-gap planar graphs [38] with *n* vertices can have up to 5n - 10 edges. Since *n*-vertex 2⁺-real face graphs have at most 4n - 8 edges (Theorem 2), none of the three families above is included in the family of 2⁺-real face graphs (and thus in the family of k^+ -real face graphs, for any $k \ge 3$). Theorem 10 proves the other direction.



FIGURE 16. A fan-planar drawing of $K_{4,b}$. The two vertex sets of the partition have distinct colors (black and white). If $b \ge 13$, this graph is not 1⁺-real face, as its crossing number is too high with respect to the maximum number of crossings allowed in any 1⁺-real face drawing.

VIII. CONCLUSION

This paper contributed to enhance the field of graph drawing beyond planarity by introducing the hierarchy of k^+ -real face graphs, for $k \ge 1$. To the best of our knowledge, this is the first beyond-planar graph family that concentrates on the properties of face-boundaries, rather than just on the crossing configurations along edges.

For any positive integer k, we provided tight upper bounds on the number of edges that a k^+ -real face graph can have, also in a setting in which all vertices must lie on the external boundary of the drawing. The edge density results have also been used to characterize the complete graphs that admit a k^+ -real face drawing, for any value $k \ge 1$, and to provide a clear picture for almost all complete bipartite graphs. We finally established an array of incomparability relationships between k^+ -real face graphs and several types of popular beyond-planar graph families. These relationships are proved as a consequence of a general result, showing that there cannot exist any beyond-planar graph class with hereditary property that includes all k^+ -real face graphs, regardless of the value of k.

A. LIMITATIONS AND OPEN PROBLEMS

The contribution of this paper mainly focused on providing edge density bounds and relationships with previous graph classes in the field of beyond-planar graph drawing. Our combinatorial findings leave some unanswered questions; also, several algorithmic aspects remain mostly unexplored.

Notation	Description
G	Connected graph
Г	Drawing of a connected graph G
V(G)	Set of vertices of a graph G
E(G)	Set of edges of a graph G
$V(\Gamma)$	Set of vertices of a drawing Γ
$E(\Gamma)$	Set of edges of a drawing Γ
$F(\Gamma)$	Set of faces of a drawing Γ
$F_{\rm int}(\Gamma)$	Set of internal faces of a drawing Γ
n	Number of vertices of G, i.e., $n = V(G) $
m	Number of edges of G, i.e., $m = E(G) $
ν	Number of vertices of Γ , i.e., $\nu = V(\Gamma) $
μ	Number of edges of Γ , i.e., $m = E(\Gamma) $
φ	Number of faces of Γ , i.e., $\varphi = F(\Gamma) $
φ_{int}	Number of internal faces of Γ , i.e., $\varphi = F_{int}(\Gamma) $
χ	Number of crossing-vertices of Γ , i.e., $\chi = \nu - n$
$\delta_G(v)$	Degree of a vertex v in a graph G
$\delta_{\Gamma}(v)$	Degree of a vertex v in a drawing Γ
$\delta_{\Gamma}(f)$	Degree of a face f in a drawing Γ
$\delta_{\Gamma}^{r}(f)$	Real-vertex degree of a face f in a drawing Γ
$\delta^c_{\Gamma}(f)$	Crossing-vertex degree of a face f in a drawing Γ

TABLE 5. Glossary of the main symbols used in the paper.

In the following we list some research directions that we consider among the most relevant for future advances.

• Research direction 1. As we have seen, both optimal 2-planar graphs and optimal 1⁺-real face graphs have density 5n - 10. Also, Theorem 10 implies that there are 1⁺-real face graphs that are not 2-planar. It remains open to establish whether the family of 2-planar graphs and the family of 1^+ -real face graphs are incomparable or not. Specifically, can we find a 2-planar graph that is not 1⁺-real face? Observe that every 2-planar drawing of an optimal 2-planar graph G is not 1^+ -real face, as it contains 0-real faces. However, one cannot exclude in principle that G admits a 1^+ -real face drawing that is not 2-planar. In the same direction, one can wonder whether there exist incomparability or inclusion relationships between the family of 1⁺-real face graphs and the families of min-2-planar, fan-planar, and 1-gap planar graphs, all containing *n*-vertex graphs with up to 5n - 110 edges (see Table 2). In fact, at least for fan-planar graphs, the results that we established in Section VI-B for complete bipartite graphs lead to the following.

Lemma 7: There exist infinitely many fan-planar graphs that are not 1^+ -real face. In other words, the families of 1^+ -real face and fan-planar graphs are incomparable.

Proof: As observed in Section VI-B, each graph in the infinite family of $K_{4,b}$ for $b \ge 13$ is not 1⁺-real face. However, all the graphs in this family can be easily drawn as fan-planar graphs, as shown in Figure 16. \Box

• Research direction 2. A second line of research is to establish the complexity of testing whether a graph is k^+ -real face or outer k^+ -real face for a given k. We remark that this type of decision problems is

known to be NP-hard for most of the families in beyond-planar graph drawing, while it is sometimes solvable in polynomial time when we restrict to optimal graphs or other subfamilies (see, e.g., [30], [49], [81]). If the recognition problem turns out to be NP-hard also for the k^+ -real face graphs, one can investigate its parameterized complexity. A related work in this direction is [82]. We observe that our results lead to some preliminary consequences about the recognition of optimal k^+ -real face graphs when k > 2. Namely, by Theorem 1, recognizing optimal k^+ -real face graphs, for $k \ge 3$, corresponds to recognizing planar graphs that have an embedding with all faces of degree k. This last problem is polynomial-time solvable for $k \leq k$ 6 and it is NP-complete for odd k > 7 and for even $k \geq 10$ [83]. Further, by Theorem 2, recognizing optimal 2^+ -real face graphs coincides with the problem of recognizing optimal 1-planar graphs, which is lineartime solvable [49].

• Research direction 3. It is interesting to study geometric k^+ -real face drawings, that is, k^+ -real face drawings with straight-line edges. For example, as already done for other beyond-planar graph families [39], [63], [71], [72], [73], [74], [84] one can either establish tight upper bounds on the edge density of graphs that admit geometric k^+ -real face drawings or characterize the non-planar embeddings that admit such a drawing. As a preliminary consideration, we observe that the geometric setting for k^+ -real face drawings is actually more restrictive than the topological setting. For example, our Theorem 2 proves that every 2^+ -real face drawing of graph with 4n - 8 edges is also 1-planar. On the other hand, geometric 1-planar drawings have at most 4n - 9 edges [63]. This implies that also the density of geometric 2⁺-real face drawings is at most 4n - 9.

Research direction 4. As explained in the introduction of this paper, one of the practical motivations behind the study of beyond-planar graph families is to increase the readability of non-planar graph layouts. Nonetheless, user studies that assess the advantage of avoiding specific crossing configurations only exist for a limited number of beyond-planar graph families (see, e.g., [15], [45], [85], [86]). It would be interesting to design human cognitive experiments aimed to assess users' analytical capabilities when k^+ -real face drawings are adopted in place of drawings that contain several 0-real faces. Additionally, for the realization of practical visualization systems, it is fundamental the design and experimentation of efficient algorithms that compute k^+ -real face drawings or approximations of them.

ACKNOWLEDGMENT

An earlier version of this paper appears in the Proceedings of the 49th International Workshop on Graph-Theoretic Concepts in Computer Science (WG 2023) [1]. In addition to the content of [1], the present article: 1) contains full proofs and technical details that were omitted or only sketched in [1]; 2) it contains a related work section; 3) it contains a new section with results on complete graphs and complete bipartite graphs; 4) it improves and extends the introduction and the conclusions to better motivate the subject of the paper and to highlight future research directions that can be of interest for the scientific community; 5) it extends the presentation of the results about inclusion relationships; and 6) it contains additional and revised figures, and more than references in the bibliography.

REFERENCES

- [1] C. Binucci, G. Di Battista, W. Didimo, S. Hong, M. Kaufmann, G. Liotta, P. Morin, and A. Tappini, "Nonplanar graph drawings with K vertices per face," in *Graph-Theoretic Concepts in Computer Science* (Lecture Notes in Computer Science), vol. 14093, D. Paulusma and B. Ries, Eds. Berlin, Germany: Springer, 2023, pp. 86–100.
- [2] R. Tamassia, Handbook on Graph Drawing and Visualization. Boca Raton, FL, USA: CRC Press, 2013. [Online]. Available: https://www.crcpress.com/Handbook-of-Graph-Drawing-and-Visualization/Tamassia/9781584884125
- [3] G. Di Battista, P. Eades, R. Tamassia, and I. G. Tollis, *Graph Drawing: Algorithms for the Visualization of Graphs*. Upper Saddle River, NJ, USA: Prentice-Hall, 1999.
- [4] M. Kaufmann and D. Wagner, *Drawing Graphs: Methods and Models* (Lecture Notes in Computer Science), vol. 2025. Berlin, Germany: Springer, 2001, doi: 10.1007/3-540-44969-8.
- [5] C. Binucci, U. Brandes, T. Dwyer, M. Gronemann, R. von Hanxleden, M. J. van Kreveld, P. Mutzel, M. Schaefer, F. Schreiber, and B. Speckmann, "10 reasons to get interested in graph drawing," in *Computing and Software Science* (Lecture Notes in Computer Science), vol. 10000, B. Steffen and G. J. Woeginger, Eds. Berlin, Germany: Springer, 2019, pp. 85–104.
- [6] L. Consalvi, W. Didimo, G. Liotta, and F. Montecchiani, "In-browser visualization of large-scale graphs," in *Proc. 1st Italian Conf. Big Data Data Sci.*, vol. 3340, M. Anisetti, A. Bonifati, N. Bena, C. A. Ardagna, and D. Malerba, Eds., Milan, Italy, Sep. 2022, pp. 27–38. [Online]. Available: https://ceur-ws.org/Vol-3340/paper29.pdf

- [7] W. Didimo and G. Liotta, Graph Visualization and Data Mining. Hoboken, NJ, USA: Wiley, 2006, ch. 3, pp. 35–63, doi: 10.1002/9780470073049.ch3.
- [8] W. Didimo, L. Grilli, G. Liotta, L. Menconi, F. Montecchiani, and D. Pagliuca, "Combining network visualization and data mining for tax risk assessment," *IEEE Access*, vol. 8, pp. 16073–16086, 2020, doi: 10.1109/ACCESS.2020.2967974.
- [9] T. von Landesberger, A. Kuijper, T. Schreck, J. Kohlhammer, J. J. van Wijk, J. Fekete, and D. W. Fellner, "Visual analysis of large graphs: State-of-theart and future research challenges," *Comput. Graph. Forum*, vol. 30, no. 6, pp. 1719–1749, Sep. 2011, doi: 10.1111/j.1467-8659.2011.01898.x.
- [10] L. Giovannangeli, F. Lalanne, D. Auber, R. Giot, and R. Bourqui, "Toward efficient deep learning for graph drawing (DL4GD)," *IEEE Trans. Vis. Comput. Graph.*, vol. 30, no. 2, pp. 1516–1532, Feb. 2024, doi: 10.1109/TVCG.2022.3222186.
- [11] B. La Rosa, G. Blasilli, R. Bourqui, D. Auber, G. Santucci, R. Capobianco, E. Bertini, R. Giot, and M. Angelini, "State of the art of visual analytics for explainable deep learning," *Comput. Graph. Forum*, vol. 42, no. 1, pp. 319–355, Feb. 2023, doi: 10.1111/cgf.14733.
- [12] R. Nararatwong, N. Kertkeidkachorn, and R. Ichise, "Knowledge graph visualization: Challenges, framework, and implementation," in *Proc. IEEE 3rd Int. Conf. Artif. Intell. Knowl. Eng. (AIKE)*, Laguna Hills, CA, USA, Dec. 2020, pp. 174–178, doi: 10.1109/AIKE48582.2020.00034.
- [13] O.-H. Kwon, T. Crnovrsanin, and K.-L. Ma, "What would a graph look like in this layout? A machine learning approach to large graph visualization," *IEEE Trans. Vis. Comput. Graphics*, vol. 24, no. 1, pp. 478–488, Jan. 2018, doi: 10.1109/TVCG.2017.2743858.
- [14] J. Pach, "Geometric graph theory," in *Handbook of Discrete and Computational Geometry*, 2nd ed. Boca Raton, FL, USA: CRC Press, 2004, pp. 219–238.
- [15] M. Burch, W. Huang, M. Wakefield, H. C. Purchase, D. Weiskopf, and J. Hua, "The state of the art in empirical user evaluation of graph visualizations," *IEEE Access*, vol. 9, pp. 4173–4198, 2021, doi: 10.1109/ACCESS.2020.3047616.
- [16] W. Didimo, G. Liotta, and F. Montecchiani, "A survey on graph drawing beyond planarity," ACM Comput. Surv., vol. 52, no. 1, pp. 1–37, Jan. 2020.
- [17] S. Hong, "Beyond planar graphs: Introduction," in *Beyond Planar Graphs*. Berlin, Germany: Springer, 2020, pp. 1–9.
- [18] S. Hong, M. Kaufmann, S. G. Kobourov, and J. Pach, "Beyond-planar graphs: Algorithmics and combinatorics (Dagstuhl seminar 16452)," *Dagstuhl Rep.*, vol. 6, no. 11, pp. 35–62, 2016.
- [19] S. G. Kobourov, G. Liotta, and F. Montecchiani, "An annotated bibliography on 1-planarity," *Comput. Sci. Rev.*, vol. 25, pp. 49–67, 2017.
- [20] J. Pach and G. Toth, "Graphs drawn with few crossings per edge," *Combinatorica*, vol. 17, no. 3, pp. 427–439, Sep. 1997.
- [21] E. Ackerman, "On the maximum number of edges in topological graphs with no four pairwise crossing edges," *Discrete Comput. Geometry*, vol. 41, no. 3, pp. 365–375, Apr. 2009.
- [22] E. Ackerman and G. Tardos, "On the maximum number of edges in quasiplanar graphs," *J. Combinat. Theory, Ser. A*, vol. 114, no. 3, pp. 563–571, 2007.
- [23] N. Alon and P. Erdos, "Disjoint edges in geometric graphs," Discrete Comput. Geometry, vol. 4, pp. 287–290, Aug. 1989.
- [24] J. Pach and J. Torocsik, "Some geometric applications of Dilworth's theorem," *Discr. Comput. Geom.*, vol. 12, pp. 1–7, Jul. 1994.
- [25] A. Suk and B. Walczak, "New bounds on the maximum number of edges in k-quasi-planar graphs," in *Computational Geometry* (Lecture Notes in Computer Science), vol. 8242, S. K. Wismath and A. Wolff, Eds. Berlin, Germany: Springer, 2013, pp. 95–106.
- [26] A. Suk and B. Walczak, "New bounds on the maximum number of edges in k-quasi-planar graphs," *Comput. Geom.*, vol. 50, pp. 24–33, Jan. 2015.
- [27] C. Binucci, A. Bungener, G. Di Battista, W. Didimo, V. Dujmovic, S. Hong, M. Kaufmann, G. Liotta, P. Morin, and A. Tappini, "Min-k-planar drawings of graphs," 2023, arXiv:2308.13401.
- [28] C. Binucci, A. Bungener, G. Di Battista, W. Didimo, V. Dujmovic, S. Hong, M. Kaufmann, G. Liotta, P. Morin, and A. Tappini, "Min-k-planar drawings of graphs," in *Graph Drawing* (Lecture Notes in Computer Science). Berlin, Germany: Springer, 2023.
- [29] M. A. Bekos, S. Cornelsen, L. Grilli, S. Hong, and M. Kaufmann, "On the recognition of fan-planar and maximal outer-fan-planar graphs," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 8871, C. A. Duncan and A. Symvonis, Eds. Berlin, Germany: Springer, 2014, pp. 198–209.

- [30] M. A. Bekos, S. Cornelsen, L. Grilli, S. Hong, and M. Kaufmann, "On the recognition of fan-planar and maximal outer-fan-planar graphs," *Algorithmica*, vol. 79, no. 2, pp. 401–427, 2017.
- [31] C. Binucci, M. Chimani, W. Didimo, M. Gronemann, K. Klein, J. Kratochvil, F. Montecchiani, and I. G. Tollis, "2-layer fan-planarity: From caterpillar to stegosaurus," in *Graph Drawing and Network Visualization* (Lecture Notes in Computer Science), vol. 9411. Berlin, Germany: Springer, 2015, pp. 281–294.
- [32] C. Binucci, M. Chimani, W. Didimo, M. Gronemann, K. Klein, J. Kratochvíl, F. Montecchiani, and I. G. Tollis, "Algorithms and characterizations for 2-Layer fan-planarity: From caterpillar to stegosaurus," *J. Graph Algorithms Appl.*, vol. 21, no. 1, pp. 81–102, 2017.
- [33] C. Binucci, E. Di Giacomo, W. Didimo, F. Montecchiani, M. Patrignani, A. Symvonis, and I. G. Tollis, "Fan-planarity: Properties and complexity," *Theor. Comput. Sci.*, vol. 589, pp. 76–86, Jul. 2015.
- [34] C. Binucci, E. Di Giacomo, W. Didimo, F. Montecchiani, M. Patrignani, and I. G. Tollis, "Fan-planar graphs: Combinatorial properties and complexity results," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 8871, C. A. Duncan and A. Symvonis, Eds. Berlin, Germany: Springer, 2014, pp. 186–197.
- [35] M. Kaufmann and T. Ueckerdt, "The density of fan-planar graphs," *Electron. J. Combinatorics*, vol. 29, no. 1, pp. 1–25, Feb. 2022, doi: 10.37236/10521.
- [36] C. Bachmaier, I. Rutter, and P. Stumpf, "1-gap planarity of complete bipartite graphs," in *Graph Drawing and Network Visualization* (Lecture Notes in Computer Science), vol. 11282, T. C. Biedl and A. Kerren, Eds. Berlin, Germany: Springer, 2018, pp. 646–648.
- [37] S. W. Bae, J. Baffier, J. Chun, P. Eades, K. Eickmeyer, L. Grilli, S. Hong, M. Korman, F. Montecchiani, I. Rutter, and C. D. Töth, "Gapplanar graphs," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 10692, F. Frati and K. Ma, Eds. Berlin, Germany: Springer, 2017, pp. 531–545.
- [38] S. W. Bae, J. Baffier, J. Chun, P. Eades, K. Eickmeyer, L. Grilli, S. Hong, M. Korman, F. Montecchiani, I. Rutter, and C. D. Töth, "Gap-planar graphs," *Theor. Comput. Sci.*, vol. 745, pp. 36–52, Jan. 2018.
- [39] W. Didimo, "Right angle crossing drawings of graphs," in *Beyond Planar Graphs*. Berlin, Germany: Springer, 2020, pp. 149–169.
- [40] W. Didimo, P. Eades, and G. Liotta, "Drawing graphs with right angle crossings," *Theor. Comput. Sci.*, vol. 412, no. 39, pp. 5156–5166, Sep. 2011.
- [41] E. Ackerman, R. Fulek, and C. D. Tóth, "On the size of graphs that admit polyline drawings with few bends and crossing angles," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 6502. Berlin, Germany: Springer, 2010, pp. 1–12.
- [42] E. Di Giacomo, W. Didimo, G. Liotta, and H. Meijer, "Area, curve complexity, and crossing resolution of non-planar graph drawings," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 5849, D. Eppstein and E. R. Gansner, Eds. Berlin, Germany: Springer, 2009, pp. 15–20.
- [43] E. Di Giacomo, W. Didimo, G. Liotta, and H. Meijer, "Area, curve complexity, and crossing resolution of non-planar graph drawings," *Theory Comput. Syst.*, vol. 49, no. 3, pp. 565–575, Oct. 2011.
- [44] V. Dujmovic, J. Gudmundsson, P. Morin, and T. Wolle, "Notes on large angle crossing graphs," *Chicago J. Theor. Comput. Sci.*, vol. 2011, no. 4, 2011.
- [45] W. Huang, P. Eades, and S.-H. Hong, "Larger crossing angles make graphs easier to read," *J. Vis. Lang. Comput.*, vol. 25, no. 4, pp. 452–465, Aug. 2014, doi: 10.1016/j.jvlc.2014.03.001.
- [46] S. Avital and H. Hanani, "Graphs," Gilyonot Lematematika, vol. 3, pp. 2–8, Jan. 1966.
- [47] B. Bollobás, *Extremal Graph Theory*. New York, NY, USA: Academic Press, 1978.
- [48] Y. S. Kupitz, *Extremal Problems in Combinatorial Geometry* (Lecture Notes in Computer Science). Aarhus, Denmark: Aarhus Univ. 1979.
- [49] F. J. Brandenburg, "Recognizing optimal 1-planar graphs in linear time," *Algorithmica*, vol. 80, no. 1, pp. 1–28, Jan. 2018.
- [50] F. J. Brandenburg, W. Didimo, W. S. Evans, P. Kindermann, G. Liotta, and F. Montecchiani, "Recognizing and drawing IC-planar graphs," *Theor. Comput. Sci.*, vol. 636, pp. 1–16, Jul. 2016.
- [51] S.-H. Hong, P. Eades, N. Katoh, G. Liotta, P. Schweitzer, and Y. Suzuki, "A linear-time algorithm for testing outer-1-planarity," *Algorithmica*, vol. 72, no. 4, pp. 1033–1054, Aug. 2015.
- [52] J. Pach and G. Toth, "Graphs drawn with few crossings per edge," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 1190, S. C. North, Ed. Berlin, Germany: Springer, 1996, pp. 345–354.

- [53] M. A. Bekos, E. Di Giacomo, W. Didimo, G. Liotta, F. Montecchiani, and C. Raftopoulou, "Edge partitions of optimal 2-plane and 3-plane graphs," *Discrete Math.*, vol. 342, no. 4, pp. 1038–1047, Apr. 2019.
- [54] J. Fujisawa, K. Segawa, and Y. Suzuki, "The matching extendability of optimal 1-Planar graphs," *Graphs Combinatorics*, vol. 34, no. 5, pp. 1089–1099, Sep. 2018.
- [55] K. Sone and Y. Suzuki, "Optimal 1-planar multigraphs," Discrete Math., vol. 346, no. 10, Oct. 2023, Art. no. 113553.
- [56] Y. Suzuki, "Optimal 1-planar graphs which triangulate other surfaces," Discrete Math., vol. 310, no. 1, pp. 6–11, Jan. 2010.
- [57] Y. Suzuki, "K₇-minors in optimal 1-planar graphs," *Discret. Math.*, vol. 340, no. 6, pp. 1227–1234, 2017.
- [58] P. Angelini, M. A. Bekos, F. J. Brandenburg, G. Da Lozzo, G. Di Battista, W. Didimo, M. Hoffmann, G. Liotta, F. Montecchiani, I. Rutter, and C. D. Tóth, "Simple k-planar graphs are simple (k+1)-quasiplanar," *J. Combinat. Theory, Ser. B*, vol. 142, pp. 1–35, May 2020.
- [59] P. Ali, P. Dankelmann, and S. Mukwembi, "The radius of k-connected planar graphs with bounded faces," *Discrete Math.*, vol. 312, no. 24, pp. 3636–3642, Dec. 2012.
- [60] B. Du Preez, "Plane graphs with large faces and small diameter," Australas. J. Combinatorics, vol. 80, no. 3, pp. 401–418, 2021.
- [61] Y. Lan, Y. Shi, and Z. Song, "Extremal H-free planar graphs," *Electron. J. Combinatorics*, vol. 26, no. 2, p. 2, 2019.
- [62] P. Bose, D. Kirkpatrick, and Z. Li, "Worst-case-optimal algorithms for guarding planar graphs and polyhedral surfaces," *Comput. Geometry*, vol. 26, no. 3, pp. 209–219, Nov. 2003.
- [63] W. Didimo, "Density of straight-line 1-planar graph drawings," Inf. Process. Lett., vol. 113, no. 7, pp. 236–240, Apr. 2013.
- [64] S. Hong, P. Eades, N. Katoh, G. Liotta, P. Schweitzer, and Y. Suzuki, "A linear-time algorithm for testing outer-1-planarity," in *Graph Drawing* (Lecture Notes in Computer Science), vol. 8242, S. K. Wismath and A. Wolff, Eds. Berlin, Germany: Springer, 2013, pp. 71–82, doi: 10.1007/978-3-319-03841-4_7.
- [65] M. Kaufmann, B. Klemz, K. Knorr, M. M. Reddy, F. Schroder, and T. Ueckerdt, "The density formula: One lemma to bound them all," 2023, arXiv:2311.06193.
- [66] J. Pach, R. Radoicic, G. Tardos, and G. Toth, "Improving the crossing lemma by finding more crossings in sparse graphs," *Discrete Comput. Geometry*, vol. 36, no. 4, pp. 527–552, Dec. 2006.
- [67] E. Ackerman, "On topological graphs with at most four crossings per edge," *Comput. Geometry*, vol. 85, Dec. 2019, Art. no. 101574.
- [68] P. Eades and G. Liotta, "Right angle crossing graphs and 1-planarity," *Discrete Appl. Math.*, vol. 161, nos. 7–8, pp. 961–969, May 2013, doi: 10.1016/j.dam.2012.11.019.
- [69] P. Hall, "On representatives of subsets," J. London Math. Soc., vol. 1, no. 1, pp. 26–30, Jan. 1935.
- [70] I. Fáry, "On straight-line representation of planar graphs," Acta Sci. Math., vol. 11, pp. 229–233, Jan. 1948.
- [71] P. Eades, S. Hong, G. Liotta, N. Katoh, and S. Poon, "Straight-line drawability of a planar graph plus an edge," in *Algorithms and Data Structures* (Lecture Notes in Computer Science), vol. 9214, F. Dehne, J. Sack, and U. Stege, Eds. Berlin, Germany: Springer, 2015, pp. 301–313, doi: 10.1007/978-3-319-21840-3_25.
- [72] S.-H. Hong, P. Eades, G. Liotta, and S.-H. Poon, "Fáry's theorem for 1-planar graphs," in *COCOON* (Lecture Notes in Computer Science), vol. 7434. Berlin, Germany: Springer, 2012, pp. 335–346.
- [73] S.-H. Hong and H. Nagamochi, "Re-embedding a 1-plane graph for a straight-line drawing in linear time," *Theor. Comput. Sci.*, vol. 892, pp. 132–154, Nov. 2021, doi: 10.1016/j.tcs.2021.09.015.
- [74] C. Thomassen, "Rectilinear drawings of graphs," J. Graph Theory, vol. 12, no. 3, pp. 335–341, Sep. 1988.
- [75] Y. Suzuki, "1-planar graphs," in *Beyond Planar Graphs, Communications of NII Shonan Meetings*, S. Hong and T. Tokuyama, Eds. Berlin, Germany: Springer, 2020, pp. 47–68.
- [76] M. A. Bekos, M. Kaufmann, and C. N. Raftopoulou, "On optimal 2- and 3-planar graphs," in *Proc. SoCG*, vol. 77, 2017, pp. 1–16.
- [77] C. Auer, C. Bachmaier, F. J. Brandenburg, A. Gleißner, K. Hanauer, D. Neuwirth, and J. Reislhuber, "Outer 1-planar graphs," *Algorithmica*, vol. 74, no. 4, pp. 1293–1320, Apr. 2016.
- [78] C. Zarankiewicz, "On a problem of P. Turán concerning graphs," *Fundamenta Mathematicae*, vol. 41, no. 1, pp. 137–145, 1955.
- [79] D. Kleitman, "The crossing number of K_{5,n}," *J. Combinat. Theory*, vol. 9, pp. 315–323, Dec. 1970.

- [80] D. R. Woodall, "Cyclic-order graphs and Zarankiewicz's crossing-number conjecture," J. Graph Theory, vol. 17, no. 6, pp. 657–671, Dec. 1993.
- [81] S.-H. Hong and H. Nagamochi, "A linear-time algorithm for testing full outer-2-planarity," *Discrete Appl. Math.*, vol. 255, pp. 234–257, Feb. 2019, doi: 10.1016/j.dam.2018.08.018.
- [82] M. J. Bannister, S. Cabello, and D. Eppstein, "Parameterized complexity of 1-Planarity," J. Graph Algorithms Appl., vol. 22, no. 1, pp. 23–49, 2018.
- [83] G. Da Lozzo, V. Jelinek, J. Kratochvil, and I. Rutter, "Planar embeddings with small and uniform faces," in *Algorithms and Computation* (Lecture Notes in Computer Science), vol. 8889, H. Ahn and C. Shin, Eds. Berlin, Germany: Springer, 2014, pp. 633–645, doi: 10.1007/978-3-319-13075-0_50.
- [84] E. Ackerman, "A note on 1-planar graphs," Discrete Appl. Math., vol. 175, pp. 104–108, Oct. 2014.
- [85] C. Binucci, H. Förster, J. Katheder, and A. Tappini, "Evaluating animation parameters for morphing edge drawings," in *Graph Drawing and Network Visualization* (Lecture Notes in Computer Science), vol. 14465, M. A. Bekos and M. Chimani, Eds. Berlin, Germany: Springer, 2023, pp. 246–262, doi: 10.1007/978-3-031-49272-3_17.
- [86] C. Binucci, G. Liotta, F. Montecchiani, and A. Tappini, "Partial edge drawing: Homogeneity is more important than crossings and ink," in *Proc. 7th Int. Conf. Inf., Intell., Syst. Appl. (IISA)*, Chalkidiki, Greece, Jul. 2016, pp. 1–6, doi: 10.1109/IISA.2016.7785427.



CARLA BINUCCI received the Ph.D. degree in computer engineering from the University of Perugia, in 2003. She is currently an Assistant Professor of computer science with the University of Perugia. She has participated in program and organizing committees for various international conferences. She has been involved in several research projects, including initiatives supported by European community. Her research interests include graph drawing, network visualization, and algorithm design.



GIUSEPPE DI BATTISTA received the Ph.D. degree in computer science from the University of Rome "La Sapienza." He is currently a Professor of computer science and engineering and a Faculty Member with the Department of Engineering, Roma Tre University. His current research interests include graph drawing, information visualization, and computer networks.







VIDA DUJMOVIĆ received the Ph.D. degree in computer science from McGill University, in 2004. She is currently a Professor of computer science and the University Research Chair with the University of Ottawa, Canada. Her current research interests include graph theory and discrete geometry.



SEOK-HEE HONG received the Ph.D. degree in computer science and engineering from Ewha Womans University, South Korea. She is currently a Professor with The University of Sydney. Her research interests include graph drawing, algorithm, network visualization, and visual analytics. She serves as a Steering Committee Member of the International Symposium on Algorithms and Computations (ISAAC) and the IEEE Pacific Visualization Symposium (PacificVis).



MICHAEL KAUFMANN received the Ph.D. degree in computer science from Saarland University, Saarbrücken. Since 1993, he has been leading the Algorithmics Group, University of Tübingen, Germany. His research interests include combinatorial algorithms, graph theory and visualization, and aspects of combinatorics and geometry.



GIUSEPPE LIOTTA (Senior Member, IEEE) received the Ph.D. degree from the Sapienza University of Rome. He is currently a Professor of computer science with the University of Perugia. On these topics he edited special issues, wrote surveys papers, book chapters, and published about 300 research articles. His research interests include algorithm design, computational geometry, network visualization, and graph drawing. He currently serves as the Editor-in-Chief for *Computer Science Review*.



PAT MORIN received the Ph.D. degree in computer science from Carleton University, in 2001. He is currently a Professor of computer science with Carleton University. His current research interests include data structures, algorithms, and graph theory.



ALESSANDRA TAPPINI received the Ph.D. degree in information engineering from the University of Perugia, in 2020. She is currently a Postdoctoral Researcher with the University of Perugia. During the Ph.D. program, she spent some months with the Algorithmics Group, University of Tübingen. Her research interests include graph drawing, graph algorithms, and network visualization. She has been a PC Member of the 30th Symposium on Graph Drawing and Network Visualization (GD 2022).

...

Open Access funding provided by 'Università degli Studi di Perugia' within the CRUI CARE Agreement